

Date of publication xxxx 00, 0000, date of current version xxxx 00, 0000.

Digital Object Identifier 10.1109/ACCESS.2017.Doi Number

Simultaneous Wireless Information and Power Transfer with Cooperative Relaying for Next-Generation Wireless Networks: A Review

Nausheen Ashraf¹, Shahzad Amin Sheikh¹, Sajjad Ahmad Khan², Ibraheem Shayea², Marium Jalal³

¹Department of Electrical Engineering, CEME, National University of Science and Technology (NUST), Islamabad, Pakistan ²Department of Electronics and Communication Engineering, Istanbul Technical University (ITU), Istanbul, Turkey ³Department of Electronic Engineering, Fatima Jinnah Women University, Rawalpindi, Pakistan

Corresponding author: Nausheen Ashraf (naushraf.ee17ceme@student.nust.edu.pk).

This study is carried out with the collaboration of the Istanbul Technical University (ITU), Turkey, supported by a TUBITAK Project No. 118C276.

ABSTRACT Wireless Power Transfer (WPT) is an innovative technology employed for enhancing the energy sustainability of wireless devices with a limited life span. The idea of integrating WPT in wireless communication leads to the idea of Simultaneous Wireless Information and Power Transfer (SWIPT) that transfers information and power to wireless devices simultaneously, thereby resulting in a drastic increase in spectral efficiency of the network. SWIPT aided Cooperative Relaying (CoR) has emerged as a new trend for Fifth Generation (5G) and Beyond 5G (B5G) systems owing to the rapidly increasing challenges faced by these networks. Cooperative relaying combined with SWIPT can be helpful in overcoming the rising demands of next generation wireless networks by providing an enhanced date rate, low latency, shorter coverage, wide spread connectivity of massive number of devices along with energy-efficiency. This article provides a comprehensive review of SWIPT technology that enables the use of CoR networks for 5G and B5G mobile networks including the significance, technologies, and protocols which can be applied. This article also examines the deployment of cooperative SWIPT involving a single relay, multiple relays and optimal relay selection, multi antenna systems and optimal beamforming .SWIPT under the influence of Hardware Impairments (HI), imperfect Channel State Information (CSI), non-linear energy harvesting models, Intelligent Reconfigurable Surface (IRS), massive MIMO, massive access for the Internet of Things (IoT) has been discussed in detail. Meanwhile, this study discusses key challenges being faced in the implementation of SWIPT for future wireless networks that need to be addressed efficiently.

INDEX TERMS Cooperative Relaying (CoR), Fifth Generation (5G), Multiple-Input-Multiple-Output (MIMO), Internet of Things (IoT), Simultaneous Wireless Information and Power Transfer (SWIPT), Wireless Power Transfer (WPT).

I. INTRODUCTION

Wireless communication has been progressing enormously for the last two decades and has become an indispensable part of our everyday lives. With the rising inclination towards smart devices aiming to support rigorous and advanced data applications, some significant challenges are in wireless communications due anticipated to environmental impact and high energy utilization constraints [1]. The Fifth Generation (5G) mobile communication networks have been started to be deployed in various countries as the next standard following the Fourth Generation (4G). Primarily, 5G technology provides higher data rates, broader coverage, greater bandwidth, reliable connectivity, a massive decrease in energy consumption, and low latency [2]. However in the course of standardization of 5G networks, it has been witnessed that there is no distinct enabling technology to support all 5G application demands. Thus researchers have already started working on Beyond 5G (B5G) systems by evading from the safety zone of 5G based solutions.

Existing 5G technology integrates with new communication technology to work together in offering communication services, such as IoT and Low Earth Satellite (LES) system. IoT technology is a smart framework involving uniquely recognizable devices capable of communicating wirelessly with each other on a massive scale via the Internet [1]-[3]. Usually, part of these wireless devices is equipped with batteries having a limited life span. With a large number of energy-constrained wireless sensors, especially those entrenched in high-risk environments, replacing batteries cannot be carried out very easily [4]. Also, in order to extend the lifetime of these small sensors by battery replacement, high cost is usually associated. Inspired by this, the primary focus is given on wireless Energy Harvesting (EH) techniques. Natural resources of energy such as solar, thermal and wind energy can be used to perform EH [5]. However, the random and impulsive nature of these resources makes it problematic to be used for applications where utmost significance is given to Quality-of-Service (QoS). Therefore, RF energy harvesting offers significant advantages as it is wireless, has low cost, and is readily accessible in the form of energy transmitted from Base Stations (BS), TV/radio broadcasting signals, and handheld radios. There are two Wireless Power Transfer (WPT) methods for energy harvesting, i.e., from ambient signals or by using a fully controlled and dedicated source of power, e.g., a BS [6].

An exceptional benefit of EH is centered at the concept of Radio Frequency (RF) signals carrying energy and information together, thus enabling the energy-deficient nodes for decoding information and scavenging energy simultaneously [7]. The two fundamental approaches, Wireless Information Transfer (WIT) and WPT, can be used for communication-only and power-only, respectively. On the other hand, Simultaneous Wireless Information and Power Transmission (SWIPT) is a combined scheme that can transfer power and carry information at the same time. It gently grows and provides conciliation between the two basic approaches and uses the RF spectrum adequately to initiate communication and energize the system. This technique facilitates and enables trillions of low-power wirelessly connected devices, e.g., IoT, to communicate and get energy anytime and anywhere [8].

Moreover, several theoretical studies consider the same signal in the literature for both information decoding and rectification [9]. However, this simultaneous transfer is not possible because of practical circuit limitations [10]. Nevertheless, the idea inspired the designing of practically realizable receivers with separate decoding and harvesting energy schemes by employing time switching or power splitting [7]. These designs are mostly implemented in recent literature [11], [12]. In addition, SWIPT technology basically provides an efficient spectral solution for energy harvesting, thereby supplying energy and exchanging information with various ultra-low-power devices supporting heterogeneous applications [13]. SWIPT could be effectively integrated to enhance throughputs and energy sustainability jointly, particularly for B5G cellular systems with small cells, e.g., Millimeter-Wave (mmWave) technologies and massive Multiple-Input Multiple-Output (MIMO) [14].

With the rapidly increasing demand for providing data services around the world, cooperative communications involving relays have been recently encouraged as an effective solution. The reliability of data transmission is greatly enhanced by employing cooperative relaying systems [15], [16]. Also, cooperative communication relaying technology minimizes path-loss and fading effects efficiently, thereby resulting in an increased coverage area and capacity of the system.

This article presents a comprehensive survey on SWIPT based CoR system. The main focus is to provide detailed information and discussion in order to showcase the importance of SWIPT, especially by using the IoT in the B5G wireless networks. Besides, future challenges and opportunities are also provided.

This article is organized as follows: Section II presents the preliminary concepts which include background, motivation, related work and contribution of this study. Section III discusses SWIPT, its techniques, and related architecture including SWIPT with non-linear EH for IoT. A discussion on Cooperative Relaying (CoR) and CoR for the Internet of Things (IoT) is provided in section IV. Section V presents SWIPT Assisted CoR, along with relaying protocols for SWIPT CoR, based on a review of different system configurations like systems with/without direct link, multi-antenna systems, multi-relay systems, MIMO with non linaer EH and imperfect CSI, SWIPT CoR

with Hi,SWIPT at 5G and B5G new frequencies. Section VI presents SWIPT assisted CoR particularly for IoT including HI for B5G IoT networks. Challenges and future work are discussed in Section VII. Section VIII concludes the article.

II. RESEARCH BACKGROUND

Cooperative relaying equipped with energy harvesting is substantially not the same as the conventional relaying schemes [17]-[19]. However, relay nodes being energylimited devices, need wireless energy harvesting to provide better performance and an increased lifetime. More specifically, when the main power supply cannot be accessed, a convenient way for relays is to use selfsustaining RF energy [20]. Fortunately, power-constrained cooperative relaying can be accompanied by RF Energy Harvesting (RF-EH), which motivates an innovative era of academic research, revolutionizing the wireless technology. Thus integration of SWIPT with CoR for next-generation wireless networks e.g Wireless Sensor Networks (WSN), mmWave, and Device IoT. to Device (D2D) communication enables spectral efficiency as well as ensures that reliability in data communication is achieved while keeping the power, complexity, and cost to a minimum.

The network size is expected to increase substantially with the rapidly-evolving B5G wireless communication systems, mainly due to Machine-to-Machine (M2M) communications as well as sensor networks employed in smart cities, and massive IoT applications. Moreover IoT is anticipated to make huge contributions in B5G systems as apart from terrestrial systems, IoT is expected to be deployed in air, underwater (UIoT) etc. Thus massive access and massive connectivity of such large number of energyhungry devices is a challenging task. As the burden on current power resources is increased traditional sources of power should be accompanied by alternative sources, specifically RF. Among RF EH techniques, SWIPT is considered to be more energy efficient increasing the life span of the existing energy-constrained wireless devices.

A. Motivation

Energy efficiency shall play a critical role in future communication systems and be considered the main design objective for 5G and B5G radio access networks [21]. Limited device life span has always been a critical aspect of modern wireless technologies [22]. Every device is required to replace its battery after every few months or years, depending upon the energy efficiency of the device and battery size.

A considerably high operational cost, as well as difficulty in replacing or recharging device batteries in several scenarios, such as wireless sensors for medical purpose, are placed inside the human body or devices embedded in a building structure, recommends a scheme that enables wireless devices to harvest energy from the adjoining environment. Various studies have discussed renewable energy sources, e.g heat, solar, wind and so on. The optimally designing resource allocation strategies by defining several objective functions and network topologies has been investigated. On the other hand, energy harvesting becomes crucial for most applications due to these resources' intermittent nature. However, RF-enabled wireless energy harvesting provides a considerable increase in throughput, has high robustness, and increased flexibility compared to other energy harvesting schemes. Thus RF energy harvesting shall be a significant building block for many popular industrial and commercial systems in the future. Fig. 1 shows various types of energy sources available for EH.

One of the critical concerns of 5G and B5G technology is to guarantee reliability in communication at low cost, reduced complexity, and power, particularly for applications including the massive IoT as well as large-scale WSNs [22]. Also, the transmission efficiency under SWIPT is improved compared to the conventional time-division multiplexing mechanism, where the transmission of power and information is separate [23], [24]. By employing SWIPT, interference present in the communication channel could potentially destroy information content in the signal and be beneficial for EH [25]. Recently, another SWIPT application is applied in relay-assisted wireless communication, which eradicates the path-loss and fading effects and increases the energy and spectral efficiency of the network. This integration of SWIPT with CoR can be drastically important for large scale wireless sensor network such as IoT [26], [27]. Fig 2 discusses the advantages of SWIPT with CoR.

Several researches has been done on and still going on in SWIPT as well as CoR systems and a number of survey articles provide an analysis of SWIPT [28]–[33] and CoR [34]–[36].Integration of SWIPT in CoR systems have also been discussed in literature [20].SWIPT aided CoR networks and their application in modern technologies, e.g., D2D communications, cognitive radio, WSN, Non-Orthogonal Multiple Access (NOMA) and IoT as well as some future challenges have been suggested in [20].

B. Related Work

In [28], Perera *et* al. presented a review on SWIPT, its architecture, and EH using SWIPT for two user Multiple-Input Single-Output (MISO), multi-user MIMO, and broadband communication systems. They extended their work in [29], where SWIPT for next-generation technologies like NOMA, massive MIMO, D2D communication, satellite communications, and IoT was presented. Resource allocation for SWIPT systems along with energy efficient EH including COMP network as well as power allocation for D2D communication was discussed in [30]. Zhang *et* al. [31] provided a basic architectural



Fig. 1: Energy Harvesting Sources



Fig. 2: Advantages of SWIPT via CoR

summary of SWIPT. They also discussed its practicality by notifying several critical aspects ranging from the link-level to system-level design, including multi-antennas, channel coding, modulation, and interference.

The study in [32] discussed MIMO systems with SWIPT enabled relaying using smart antenna schemes. They also reviewed receiver architectures for SWIPT systems. Ku et al. [33] studied several effective physical layer security techniques affecting SWIPT performance. Additionally, they suggested the employment of massive MIMO techniques to enhance SWIPT performance via numerical results. All the works from [28]-[33] discussed SWIPT and its various applications in wireless communications. Cooperative relaying systems have also been discussed in literature. The effect of relay selection for a CoR network to enhance the capacity of systems was studied in [34]. They also discussed the optimal relay assignment algorithm that provides an optimal solution for allocating relay nodes in order to enhance capacity. Authors in [35] considered industrial wireless sensor networks and relay selection techniques with a brief review of relay selection parameters. They also discussed the diversity combining techniques for relay networks. In [36], Mansourkiaie *et* al. presented a review of routing protocols in cooperative communication networks. Along with providing the taxonomies of routing protocols, they also highlighted the limitations of each. A comprehensive study of EH SWIPT systems with cooperative relaying has been presented by Hossain *et* al. in [20]. This article focused on various relay selection techniques as well as resource allocation in SWIPT aided CoR systems. The integration of SWIPT and CoR for 5G networks has been explained in addition to a few future challenges.

C. Contribution

This article reviews SWIPT aided CoR systems, for 5G and B5G networks. Table I provides a comparison of existing literature with this work. The contributions of this article include an overview of not only SWIPT, its architecture but a detailed discussion on SWIPT systems considering both linear as well as non-linear energy harvesting models involving imperfect CSI. SWIPT CoR networks in the presence of Hardware Impairments (HI), Intelligent Reconfigurable Surface (IRS), exploring SWIPT CoR at new frequencies, massive MIMO and massive access particularly for the massive IoT have been reviewed, which are key enabling technologies supporting B5G. SWIPT assisted CoR networks, their applications and future challenges have also been presented. Apart from terrestrial IoT network, Underwater IoT (UIoT) has been explored as a future challenge for B5G systems.

III. SIMULTANEOUS WIRELESS INFORMATION & POWER TRANSFER (SWIPT)

One of the modern advances in wireless energy harvesting technology is SWIPT. In SWIPT, either dedicated or ambient RF sources [37], [38] are employed for transmitting energy and information simultaneously, as depicted in Fig. 3. It provides an advantage in terms of Electrical Efficiency (EE) and Spectral Efficiency (SE). The idea of transporting information and energy together was first discussed by Varshney et al. [39]. In order to show the fundamental performance trade-off for simultaneous information and power transfer, a capacity-energy function was suggested. By employing realistic fading channels with Additive White Gaussian Noise (AWGN), Sahai et al. extended the idea by employing a non-trivial trade-off between energy harvesting and information decoding by considering a Single-Input Single-Output (SISO) system [40]. In theory, energy and information can be extracted from the same waveform in SWIPT. However, due to circuit design limitations, such as the different power sensitivity of antennas at the receiver side, its practical implementation is impossible. An essential aspect of being noted here is that energy harvesting requires

This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703, IEEE Access



TABLE I

COMPARISON OF DIFFERENT SURVEY ARTICLES WITH CURRENT STUDY

Study	SWIPT	CoR	Domain	Integrated SWIPT- CoR	Internet of Things (IoT)
Perera et al. [28]	Yes	No	MISO, MIMO, Broadband	No	No
Perera et al. [29]	Yes	No	NOMA, D2D, WSN, Cognitive Radio	No	Yes
Huang <i>et</i> al. [30]	Yes	No	Resources Allocation, Energy Efficiency, CoMP Networks.	No	No
Zhang <i>et</i> al. [31]	Yes	No	Link and System Level Design Aspects of Wireless Systems	No	No
Z. Ding <i>et</i> al. [32]	Yes	Yes	MIMO SWIPT Systems with smart antennas	No	No
Ku <i>et</i> al. [33]	Yes	Yes	EH Techniques, Scheduling, Optimization and Design Issues considering Cognitive Radio Networks and Cellular Networks.	No	Yes
Bhute <i>et</i> al. [34]	No	Yes	Basics of Relaying, Relaying Strategies in CoR, MIMO Systems and Optimal Relay Selection.	No	No
Umamaheswari <i>et</i> al. [35]	No	Yes	Relay Selection, Diversity Combining Techniques	No	No
Mansourkiaie <i>et</i> al. [41]	No	Yes	Routing Protocols and Their Review, and Limitations in Each Protocol.	No	No
Hossain <i>et</i> al. [20]	Yes	Yes	EH, SWIPT-CoR, for 5G networks with applications in D2D,VANET,WBAN,WSN, CRN, IoT.	Yes	Yes
Current Study	Yes	Yes	Review of SWIPT assisted CoR for 5G and B5G networks including massive MIMO and massive IoT. IRS for SWIPT aided CoR systems. SWIPT CoR with HI, imperfect CSI, nonlinear EH, as well as SWIPT at new frequencies. Applications of IoT e.g UIoT for B5G systems.	Yes	Massive IoT and UIoT.

This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703. IEEE Access



Fig. 3: Basic Architecture of SWIPT



Fig. 4: Separated Receiver for SWIPT

-10dBm power sensitivity, whereas information decoding requires as lower as -60dBm [42]. In order to implement SWIPT, a signal has to be splitted into two parts, one for harvesting energy and other for decoding information.

1) SWIPT techniques

The following are the two primary techniques used for the practical implementation of SWIPT.

1.1. Separated receiver

1.2. Integrated receiver

1.1. Separated receiver

There are separate antennas for both EH and ID for this architecture, working on different fading channels [43]. With the help of this architecture, EH and ID can be done concurrently and independently. For achieving a trade-off between rate of information transmission and energy harvesting, CSI and feedback from the receiver can be utilized to obtain an optimized result, as shown in Fig. 4.

1.2. Integrated receiver

In this type of architecture, there is no separate antenna for ID and EH. This architecture has three types:

1.2.1 Time Switching (TS)

In this type of integrated receiver, same antenna is employed for EH and ID. A switch is employed at the receiver to switch between EH and ID periodically. This TS receiver



Fig. 5: Time Switching (TS) Receiver for SWIPT



Fig. 6: Power Switching (PS) Receiver for SWIPT

needs precise information and energy scheduling as well as synchronization of time along with the advantage of simplicity in hardware [42]. TS receiver is shown in Fig. 5.

1.2.2 Power Splitting (PS)

In this type of SWIPT architecture, the received signal is split using a power splitter with one part of the signal for EH and other for ID. The PS receiver is shown in Fig. 6. Both EH and ID is done simultaneously by optimization of PS ratio α . However, the hardware becomes more complicated due to the power splitter as a part of the receiver circuit [43].

1.2.3 Antenna Switching (AS)

Antenna switching can be employed for SWIPT with different antennas used for ID and EH. One group of antennas performs EH, while the other group can identify the ID. Sometimes it can be used for optimizing the receiver architecture. It can be thought of as a distinct case for PS architecture, i.e., performing EH and ID concurrently, as shown in Fig. 7.



Fig. 7: Antenna Switching (AS) Receiver for SWIPT



Fig. 8: Advantages of SWIPT



Fig. 9: Internet of Things (IoT)

It has been shown in literature that SWIPT can be employed for EH considering two energy harvesting models; linear as well as non-linear EH model. In linear EH [42], the EH circuit's power conversion efficiency defined as ratio of output power to input signal power is assumed to be constant. Thus the output power (DC) and input power are independent of each other. In a linear EH model, the conversion efficiency is a constant whose values range in the interval from [0, 1] [44]. This conversion efficiency represents the ability of the RF-DC conversion circuit. The authors studied the trade-off between rate-energy regions by developing a MIMO broadcasting channel with three nodes developing a rate-energy (R-E) region and optimal beamformer following linear energy harvesting model [12]. In [45], a MISO broadcast SWIPT system was investigated, with power splitting and transmit beamforming combined. The system was designed to minimize power transmitted with a constraint on minimum value of the received Signal to Noise Ratio (SNR). Results showed that beamforming increases the amount of energy harvested for SWIPT systems. Authors in [46] considered a multicarrier system with an energy-efficient SWIPT, where user scheduling, subcarrier as well as power allocation were discussed. In [11], the authors designed a beamformer with imperfect Channels State Information (CSI), without considering the energy consumed in channel estimation. Optimization of the system was done to achieve QoS and achieve a robust design. However, existing literature studies [11], [12], [43], and [46] are based on linear EH model where power conversion efficiency is independent of input power of the harvesting circuit. Fig. 8 shows the primary advantages of SWIPT.

However, in practical wireless networks, wireless power transfer is non-linear and dependent on practical EH circuit parameters which employ diode as a non-linear component [47]. Authors in [44] designed a non-linear EH based SWIPT model, dependent upon logistic function and real data. They proved a considerable gain in terms of average total energy harvested when compared with the traditional linear model. Boshkovska *et* al. [48] utilized a non-linear EH model and developed a resource allocation scheme for harvested power maximization. Simulation results proved the substantial increase in harvested energy and robustness against imperfect CSI as compared to conventional linear EH scheme and perfect CSI [11].

2) SWIPT for Internet of Things (IoT)

An IoT network is equipped with a special type of sensors, transceivers, batteries, and control processors attached to each object to monitor their environment and transmit and capture data from the physical world. An IoT network is shown in Fig. 9. Due to the limited battery life of embedded sensors, a lot of research interest is being developed for utilizing SWIPT in order to increase their longevity and energy efficiency, thus enabling them to become self-sufficient in their operation without compromising QoS.

Wang *et* al. [49] considered a MISO SWIPT system with frequency division duplexing and developed an optimization scheme to maximize net harvested energy in the presence of imperfect CSI. In [50], Chae *et* al. presented a SWIPT system for IoT sensor networks employing K transmitter-receiver pairs for minimizing the total power transmitted with constraints on rate and energy harvesting. By employing interference decoding and hybrid beamforming, they showed that the SWIPT performance for IoT devices could be considerably improved compared to the previous works [49], [51].

The authors in [52] considered MIMO Broadcasting Channel (MIMO-BC) with a TS receiver for an IoT network, where an optimization problem was designed for maximizing EE based on maximum transmitted power and minimum harvested energy at each user terminal.





Fig. 10: SWIPT for IoT



Fig. 11: Cooperative Relaying (CoR)

In [53], Lee *et* al. discussed the allocation of resources and task scheduling for IoT networks. The main idea was to minimize the on-grid energy consumption at the access points based on three constraints: the minimum average harvested energy and data rates and the minimum task performing rates for IoT nodes by designing a centralized algorithm. For reducing the complexity of a centralized algorithm, they also devised a distributed task scheduling algorithm that aimed at assisting each IoT device to satisfy its task performing rate requirement by efficiently utilizing its harvested energy. SWIPT for IoT is shown in Fig. 10.

Outage performance considering a primary and secondary IoT system investigated by [54], by employing Time Switching (TS) based SWIPT for secondary IoT devices that harvest energy while serving as a relay to primary IoT nodes. In [55], a MIMO system with multiple energy and information receivers considering an IoT network was proposed for SWIPT. A resource allocation algorithm was designed based on maximized harvested energy at energy receivers and security at information receivers. Authors in [56] developed an innovative SWIPT algorithm for IoT by sending wireless power via unmodulated Continuous Wave (CW) and transmitting information employing a small modulated wave for decreasing the interference and increasing the power efficiency, instead of conventional PS as well as TS schemes. They developed a novel receiver design for IoT

devices, since the receiver circuit for IoT devices is required to have low complexity and low-power consumption, so in order to get maximum utilization of harvested power for various parts of the IoT devices, they suggested the use of envelope detection for information decoding, which avoids power-consuming active components e,g mixers and oscillators. The authors derived expressions for energy harvesting efficiency for power signal and determined frequency response of information signal. The analysis results were validated by both the simulation and development of experimental testbed. Huang et al. [57] proposed a scheme by considering a Distributed Antenna System (DAS) with SWIPT and PS, where EE was derived for the IoT network. By changing the transmit power of distributed antennas and PS ratio of IoT devices, EH and ID could be coordinated.

3) SWIPT with nonlinear EH for IoT:

Since the EH devices in IoT applications are small sensors having lesser hardware size requirement and reduced circuit complexity, thus in such cases, nonlinearity in EH needs to be accounted for since it may be very difficult to practically implement various energy harvesters so as to mitigate nonlinear effects. In [58] ,Kang et al. developed DPS algorithm for SWIPT by considering an ergodic fading channel and nonlinear EH model for IoT applications in the presence of two cases namely CSIR, when CSI is known at the destination only and when CSI is known at source and destination both. The fading conditions were accurately determined at the destination using downlink pilots transmitted from source. Also by employing uplink pilots transmitted from destination, CSI was estimated at the source. The main advantage of proposed scheme was that it can be readily employed after determining the energy consumption requirement for processing as well as information transmission from IoT receiver since harvested energy in IoT sensor should be utilized not only for decoding information but also for processing and transmission of information. Numerical results validate that proposed optimal as well as suboptimal schemes show significant performance improvement as compared to existing algorithms.

Since D2D communication is a promising technology for supporting IoT networks hence authors in [59] developed an algorithm to maximize the EE of all D2D links in a D2D network by the developing the optimized resource and power allocation strategy considering non-linear EH model. Results depicted that cumulative EE is much greater when D2D communication distance is short and there are more number of users and proposed scheme achieved higher EE of all D2D links as compared to the existing algorithms.

In [60] authors discussed, a resource allocation algorithm to enable secure network considering SWIPT for IoT. The algorithm was based on non-linear EH model and an optimization problem was developed for maximizing total energy transferred to the receivers. Numerical results depict

that the proposed algorithm guarantees security in communication as well as assists in a significant performance improvement in terms of the energy harvested when compared with existing schemes employing the linear EH model.

IV. COOPERATIVE RELAYING (CoR)

By considering broadcast in the wireless medium, cooperative communication is suggested [61], by which various nodes can cooperatively send signals to a destination together. Recent advancements in technology have proved that cooperative communication aids in substantial performance improvements by increasing capacity, spatial diversity, and enhanced transmission reliability [62], [63]. In a relay-based communication system, the source first transmits its data to the Relay Nodes (RNs). Each RN after utilizing some cooperative relaying protocols forwards the data to the destination. When a signal is received from RNs, the receiver decodes the data [64]. In such a way, the nodes that are out of the source's range but the AP can communicate with them, as shown in Fig. 11.

In a cooperative communication system, the relay node is not only devoted to relaying, but it also transmits its own information. This transferring of information to each other on a voluntary basis [65] is an idea also called as cooperative diversity [66]. Apart from providing network connectivity to remote nodes, relays also can adjust the distance between them to achieve electrical efficiency, thereby reducing the effects of fading in wireless networks [67], [68]. By employing multiple relays, throughput of any network can be increased [69]. By providing spatial diversity, the throughput can be enhanced in a cooperative communication network.



Fig. 12: Cooperative Relaying (CoR) Strategies

1) Cooperative relaying strategies

There are two main strategies for CoR, i.e., fixed and adaptive relaying. For fixed relaying, resources between source and relay are allocated in a deterministic way. In adaptive relaying, relays become active only when they require processing information; otherwise they remain silent [70]. Data processing can be done according to the available protocols at the relay. Different relaying protocols are mentioned in Fig. 12 and discussed in detail below:

1.1. Fixed Amplify & Forward Relaying protocol (FAF):

In this technique, relay amplifies the signal received from the transmitter and forwards it to the receiver, hence it is also named as non-regenerative relaying. AF relaying has advantages in terms of lesser complexity and simplicity, but the main drawback is amplification of noise, thereby effecting the QoS of the network [71].

1.2. Fixed Decode & Forward Relaying protocol (FDF):

In this type of relaying, the relay first decodes and then forwards received signal to the receiver. This relaying eliminates noise as was in AF but introduces computational delay and might allow errors to propagate until the destination [72].

1.3. Code & Forward relaying protocol (CF):

Unlike AF/DF schemes, where relay sends a repeated version of bits from the source, the process is done by dividing the codeword into different segments and being sent through independent fading channels. Segments are equipped with a CRC check. The relay in coded cooperation transmits incremental redundancy, which gives a large redundancy in the codeword at the receiver when combined with that sent by the source. The chances of recovering original information in the presence of errors are increased by this redundancy [73]. However, delay and processing time increases at the cost of reliability and increased accuracy. Moreover, it also reduces interference in any CoR network.

1.4. Selective Decode & Forward (SDF) relaying protocol:

The Selective Decode & Forward (SDF) relaying protocol is an adaptive relaying mechanism that eliminates the noise being amplified as well as error being propagated in DF. Here relay decodes messages from the source, and after performing an error check, it forwards to the destination only if correct decoding took place; otherwise, it remains silent [74].

1.5. Incremental relaying protocol (IR):

The incremental relaying protocol is the highest spectral efficient protocol because a feedback channel is present from the transmitter to relay. An acknowledgment is received at the relay when the message is transmitted correctly, and it does not transmit again during the transmission phase. The second transmission phase is based upon CSI of a channel having direct link between

transmitter and receiver. However, owing to the high signaling involvement, the complexity might increase [75].

1.6. Intelligent Reconfigurable Surface (IRS):

Intelligent Reconfigurable Surface (IRS) has swaved the research community recently, primarily becoming a potential candidate technology for B5G wireless networks. IRS is composed of a set of reflecting units which are reconfigurable, and their phase shifts can be controlled according to the changes in reflection from the wireless communication environment. As a matter of fact, IRS can suitably reflect RF signals transmitted from transmitters to desired directions, such that the received signal power can be effectively enhanced at the intended destinations while suppressing interference at undesired receivers [76]. Moreover, IRS is actually an antenna array with passive elements such that the spectral as well as EE of a wireless network can be increased in a cost effective way. IRS scheme has proved to be very striking from consumption of energy point of view, mainly because it can amplify and forward incoming RF signals without using any PA, rather by appropriately calculating phase shift of each reflecting part, there by combining each reflected signal constructively. Obviously, since no PA is employed, an IRS shall utilize lesser energy as compared to an AF relay [77]. Another advantage of IRS is that it can operate without any power source, complex encoding and decoding algorithms like other technologies e.g MIMO transmit and receive beamforming and relays. The geometrical size of antennas employed at relays is comparable with the wavelength, thus making AF relay a diffuser rather than a reflector. If the geometrical size of RIS is optimized, a large increase in link budget can be obtained as compared to relaying. Also absence of ADCs and PAs makes the IRS unaffected by noise at the receivers thus avoiding addition of noise and providing a full duplex transmission [78].

Recently IRS has been combined with energy harvesting to improve SE and EE of the network. In [79] authors proposed an IRS based MISO system with multiple users by employing SWIPT.For a multi antenna system, joint optimization of transmit beamforming at the transmitter as well as passive reflective beamforming at IRS is suggested for maximizing the received power at EH receivers with SINR and maximum transmit power constraint at the ID receiver as well as the transmitter respectively. Authors in [80], discussed an IRS assisted downlink MISO system with multiple users and presented two low complexity EE maximization algorithms for transmit power allocation of BS and IRS reflector parameters. Simulation results depicted optimal SE performance achievement and proved that IRS based communication provides a greater EE as compared to relay aided one. It was also validated that optimal operating point for EE is dependent upon number of mobile users and IRS components along with individual power consumption of IRS elements. In [81], weighted sum power was maximized in an IRS assisted SWIPT system. Simulation results proved that suggested scheme employing

IRS can significantly enhance the R-E performance for SWIPT systems.

Integration of IRS in B5G IoT is a promising area to be explored. IRS has introduced a new idea of beamforming that has been previously implemented at transmitters and receivers only but now has been shifted to a controllable environment. Moreover, IRS structures can be easily integrated in the communication environment, since their very small hardware imprint allows for their easy utilization in the IoT e.g in ceilings of buildings, doors, laptop bags, upto being incorporated in human apparel [82]. In this regard, the joint assimilation of RIS with rapidly emergent technologies, such as massive MIMO, mm waves, free space optical communication, IoT, physical layer security, energy harvesting and beamforming is a favorable and new research trend. Hence, IRS is regarded as an inspiring area to be explored for B5G wireless networks with and without energy harvesting.

1.7. CoR for the Internet of Things (IoT)

Cooperative communication is a providential technique that aims to overcome small as well as large scale fading and aids in increasing the reliability of the wireless network [57]. With the advancement in wireless technologies, the 5G of wireless networks has been focusing on developing new applications in various ways of virtual reality, IoT, D2D communications, smart cities, and smart homes [83]. IoT is a network of a large number of different objects linked together and monitored via the Internet. CoR for IoT is shown in Fig. 13. Although there is a swelling interest in the field of IoT technology, several challenges still need to be investigated. One of the most significant challenges is the limited energy of the sensor nodes comprising the IoT network [84]. In order to overcome this challenge, reliable communication protocols need to be developed that increase network life span and reduce energy consumption. Lots of research has been aimed at the use of the CoR in enhancing reliability and reduction in energy consumption of wireless networks by considering either small scale fading modeled by Rayleigh or Nakagami-m distributions, or large scale fading modeled by Log-Normal distribution [85], [86].

In [87], the authors have suggested a generalized K fading channel model that encompasses multipath fading, pathloss, and shadowing for a multi-hop IoT network. They evaluated the average symbol error probability and OP for single-hop as well as multi-hop links. They also proved that for high SNR, increasing the number of hops results in increasing communication reliability depicted by average symbol error probability and OP. For improving the network reliability in IoT, Yang *et* al. [88] designed a communication protocol to guarantee the network reliability between terminals and BS in Narrowband IoT (NB-IoT).







They proposed a relay-based cooperative communication framework where D2D communication is suggested to improve network performance, i.e., direct communication between two NB-IoT terminals in a cellular network coverage area with the least involvement of BS.

Authors in [89], proposed a novel relaying framework for energy-efficient uplink transmissions in IoT networks. They formulated a linear optimization problem which was designed to determine optimal positioning of relays along with reducing the total consumed energy while fulfilling a set of constraints, including the budget constraint that decides the maximum number of IoT relays, and the QoS constraint indicated by minimum Signal to Interference Noise Ratio (SINR) of IoT devices. In [90] Bao et al. developed an Incremental Selection Hybrid Decode-Amplify Forward (ISHDAF) model for systems with single relay. They developed relay selection algorithm dependent on Hybrid Decode-Amplify-and-Forward (HDAF) scheme considering a multi-relay system along with optimization of power allocation for IoT. They proved that suggested technique can be adopted effectively in cooperative IoT relay systems in order to get high power efficiency and better outage performance [91].

V. SWIPT Assisted CoR

CoR networks equipped with energy harvesting technology, e.g., SWIPT, are getting increased attention by researchers nowadays. This combination of SWIPT and CoR provides lots of advantages for energy limited devices and helps in improving QoS of the network [92].

Relaying protocols for SWIPT based CoR are as follows:

1. Time switching relaying (TSR) protocol:

In TSR protocol, the information block is sent from transmitter to receiver node in time T. The fraction of time for harvesting energy from the source is α , with its value ranging between $0 \le \alpha \le 1$.Here, $(1 - \alpha)T$ is utilized for transfer of information, so that $(1 - \alpha)T/2$, is used for information transfer from transmitter to relay and remaining portion, given as $(1 - \alpha)T/2$, is employed for sending

information from relay to receiver [93], [94] as shown in Fig. 14.



Fig. 14: TSR protocol for SWIPT CoR

2. Power splitting relaying (PSR) protocol:

In PSR protocol, the signal after reception is split in two ratios, ρ being employed for harvesting energy, and remaining $(1 - \rho)$ is for information transmission between sources to relay, where $0 \le \rho \le 1$. This can be clearly seen in Fig. 15.



Fig. 15: PSR protocol for SWIPT CoR

3. Hybrid protocol (TSR-PSR) protocol

Atapattu *et* al. in [95] proposed a novel hybrid protocol for EH in SWIPT based CoR networks. It consists of properties of both TSR and PSR protocol. This can be utilized for all types of networks employing relay, with results proving that it outperforms both TS as well as PS protocols [94]. The relay uses αT time for harvesting energy from source, communication between sources to relay takes place in $(1 - \alpha)T/2$ time duration. A portion ρ , of signal received is used for EH at relay where $0 \le \rho \le 1$, and the remaining portion $(1 - \rho)$ is utilized for transmission of information between S to R. The communication between relay and destination takes place over $(1 - \alpha)T/2$ time duration [96]. By putting $\alpha = 0$ or $\rho = 0$, the hybrid protocol turn out to be equal to PS or TS protocol, respectively as shown in Fig. 16.



Fig. 16: Hybrid protocol for SWIPT-CoR

A. SWIPT CoR Network without Direct Link

SWIPT CoR has been studied in detail in literature for systems without Direct Link (DL) between source and destination, as depicted in Fig. 17. Nasir et al. [90] suggested a three node network having a single source, destination and relay inbetween.AF relaying along with both TSR/PSR protocols were employed for energy harvesting. Outage Probability (OP) as well as ergodic capacity were derived as a measure of throughput in both delay limited as well as delay tolerant transmission modes. They extended their work in [97] using DF relaying instead of AF. Optimal values of throughput with different noise variance values as well as distances between source and relay as well as relay to destination were investigated. Although AF and DF relaying requires additional energy consumption due to power consumed in Power Amplifier (PA) as well as in RF circuitry [98], in analysis of [97] energy required for decoding operation is neglected and complete energy that is harvested at the relay using SWIPT was used for forwarding the data to destination.

Aissa et al. [99] discussed the PSR/TSR protocol for harvesting energy with DF relaying at the relay. Here cochannel interference was also utilized to harvest energy. Comparative results showed that PSR is better than TSR at low SNR based on throughput, ergodic capacity, and OP. Considering a three-node network, Wang et al. [100] suggested optimal harvested energy and the effect of location of relay on OP in order to explore the idea of deployment of RF EH relays. In [101], the authors studied a two-way AF-based relay network where energy harvesting was done using TSR protocol. Performance analysis of TSR protocol was made based on throughput of network. Since QoS in SWIPT with CoR is directly affected by time and power fraction, authors in [102] worked on finding optimized values of TS parameter (α) and PS ratio (ρ) to maximize the capacity using Lagrange multipliers. SWIPT with cooperative DF systems and non-linear EH model with hybrid relaying protocol was studied in [103]. OP and throughput performance were studied and it was shown that optimal values of ρ and α as well as determining optimal positioning of relay minimizes OP of systems. Furthermore, SWIPT CoR with non-linear EH, the hybrid protocol performs better than TSR as well as PSR protocols in terms OP.

B. SWIPT CoR Networks with Direct Link

A lot of research has been made in SWIPT based CoR networks with DL to increase the diversity gain and performance of communication systems as shown in Fig. 18. In [104] Pan *et* al. considered a dual hop SWIPT system where relay utilizes threshold AF/DF relaying in order to make out whether to assist in S-D direct link transmission or not. OP of system is investigated by employing PSR protocol at the relay. The authors in [105] designed a CoR network where battery enabled relay utilizes SWIPT for

charging the battery under delay limited transmission mode. This work is extended to multiple relay scenario where optimal relay is chosen based on minimum outage and minimum distance between source and relay. In [106] Hoolee *et* al. proposed a half-duplex algorithm for a point to point system employing relay being energy constrained and harvesting energy from source using PSR protocol and optimized value of PS ratio (ρ) to obtain minimum OP. Rateless coded SWIPT enabled relay systems were studied in [107].

They devised optimization problems for rate maximization considering three types of receivers, i.e., ideal, PS and TS, respectively. Also, the effect of position of relays on system's performance was discussed. A three-node DF relaying model with PSR protocol for SWIPT was considered in [108], where OP was derived in the presence of Nagakami-M fading channel. They proved that lesser OP is expected in Nagakami instead of the Raleigh fading channel.

In [109], Bai *et al.* investigated PS scheme in nonlinear EH, AF relay network with a direct link. They formulated an optimization scheme for system capacity maximization according to instantaneous CSI based on nonlinear EH model considering a logistic function. Simulation results showed that increased capacity can be attained when PS is optimized with nonlinear EH scheme instead of linear one.

C. Multi-Relays SWIPT CoR Networks

There are a lot of studies in the literature which have focused on involving multiple relays to enhance systems capacity and improve QoS. SWIPT system with multi-relays is depicted in Fig. 19. In [110] a multi-relays CoR system was considered where end to end rate was maximized by optimizing power splitting ratio (ρ) of the relays for both AF/DF relaying protocols. Authors in [111] studied an AF multi-relays network and established HPTS protocol for EH at the relays. Optimal relay is selected based upon end-end SNR. OP and throughput were the performance metrics. Performance analysis of EH multi-relays network in terms of OP with Nagakami-M fading channel is discussed in [112].

Here a direct link is present in the multi-relays scheme. The results show that increase in the number of relays, decreases the OP. An analysis of TSR/PSR protocol with AF/DF relays was presented by Perera *et* al. [113]. Throughput and OP was derived by considering delay limited transmission mode. Multi-relays having direct link between source and destination was discussed. Results show that for lower values of transmission rates, TSR protocol outperforms PSR considering OP, Bit error rate (BER), and throughput. Relay selection (RS) for battery-less and battery-equipped relays were discussed in [114] using PSR for SWIPT and AF relaying protocol at the relays. By studying OP analysis, it was observed that full order diversity K is attained in both non-EH and EH relays. Considerable performance





Fig. 17: SWIPT CoR System without Direct Link



Fig. 18: SWIPT CoR System with Direct Link

improvement is attained if EH relays are used in addition to conventional relays. Here relay selection was based upon highest end to end SNR.

Li *et* al. in [115] derived performance analysis by considering relay assisted cooperative network with multiple relays. From results it was concluded that if multi-relays are utilized for outage improvement CoR can be very helpful. In this article an ON-OFF Markov model was established to discuss stochastic nature of energy harvested.

The idea of selecting optimal relay based upon rateless codes was presented in [116]. Optimization is based upon maximum achievable rate of the network. For the relays harvest-use-store PS strategy was employed. In [117], relay is selected based upon statistical and perfect CSI. Here a dynamic power allocation based relay selection is considered. The results proved that proposed scheme gives a better performance as compared to traditional min-max algorithm. Trade-off exists between acquisition of statistical CSI involving resources vs non practical (ideal) CSI. Relay selection with SWIPT aided CoR was made based upon CSI at the relays. Results showed that a trade-off exists between energy efficiency of real nodes vs number of relays present. In [118], Kridis et al. discussed relay selection for Wireless Powered Communication Network (WPCN) based on power level of battery at the relay. Relay selection in the



Fig. 19: Multi-relays SWIPT CoR System

presence of multiple relays was dependent upon battery's status. Authors in [119] suggested relay selection considering full duplex two way communication network. PSR was used for EH and Single Relay Selection (SRS) as well as Global Relay Selection (GRS) scheme is proposed.

D. Multi-Antenna SWIPT CoR Networks

Multi-antenna systems have gained significant importance in 5G and beyond networks due to the fact that with a larger number of antennas employed at the transmitter or receiver (MIMO, MISO, SIMO, mMIMO) higher SE and EE can be achieved. It also results in improved diversity along with capacity increase by reducing multipath fading and channel interference by focusing the radiation only in expected direction and then modifying radiation with reference to the signal environments or varying traffic circumstances, a technique named as beamforming[120]. Authors in [121] considered SWIPT based multi-antenna system with DF relaying protocol and TSR at the relays. Performance metrics are throughput and outage for delay limited as well as delay tolerant transmission modes. A multi-relays multiantenna SWIPT CoR system is shown in Fig. 20. Results show that an increasing the number of antennas along with beamforming enables more energy to be harvested in case of SWIPT. To explore the advantage of multi-antennas at relays Maximum Ratio Transmit (MRT) and Maximum Ratio Combining (MRC) strategies are exploited. An optimal time/power allocation scheme to maximize EE was presented in [122] for full duplex system with relay. The EE of dual vs single antenna is compared and results prove that full duplex mode outperforms half duplex mode. In [123], a joint antenna switching and PS scheme was proposed for rate maximization with optimized values of PS ratio as well as set was determined. A comparison between direct, conventional relaying and proposed scheme showed that highest rate is achieved while using proposed strategy.

A joint resource allocation scheme for multi-antenna relay using DF and PSR protocol at the relay was proposed in [124]. An antenna clustering scheme was suggested for



Fig. 20: Multi-antenna SWIPT CoR

antennas in order to perform decoding of information and harvesting energy exclusively. A greedy clustering algorithm to minimize complexity of optimal clustering of antennas was also developed. Results depicted that by increase in the number of antennas at the relay, the spatial diversity and amount of harvested energy can be increased. Authors in [125] suggested a two way multi-antenna CoR system with single antenna at source and destination .Here BER for six relaying schemes as a function of relay antennas, PS ratio and EH time was proposed.

Results showed that optimal values of power splitting ratio and time switching factor which minimized BER depends upon number of relay antennas. Authors in [126] considered a multi-relays multi-antenna system with MRC at the destination to choose best combination of signals from relays and antennas for a given SNR. OP and throughput for both TSR/PSR and AF relaying is discussed. Results show that increase in number of relays and number of relay antennas significantly improves performance of system in terms of throughput. Zahidi *et* al. in [127] proposed a multirelays multi-antenna system having single antenna at source and multiple antennas at destination. Simulation results showed that MRT strategy at the relay performs better as compared to antenna selection in terms of throughput.

In [128] Liu *et al.* designed a MIMO AF multiple-hop wireless sensor network with both PS and TS schemes applie d at the receiver. The design included multiple relays and multiple antennas to determine a combined beamforming solution for source and relay precoders for maximizing achievable rate for WSN, based on transmit power and harvested energy constraint and ignoring hardware impairments in the transceiver circuitry. In order to find joint design of beamforming matrices a diagonalization-based optimization scheme was suggested.

E. MIMO with nonlinear EH and imperfect CSI:

In [129] authors investigated a dual hop MIMO system having multi antenna source, destination as well as single antenna relay with PSR for EH and DF for processing information, by assuming negligible energy consumption for DF operation. The energy harvester at relay was considered to be utilizing a non-linear EH model having a saturation threshold. The transmission delays and the complexity of wireless environment was accounted for by assuming imperfect CSI both at source and destination. Thus authors employed a TAS scheme at the source by considering outdated CSI fed back from the EH relay and MRC at the destination in order to handle duplicate copies of signals with channel estimation errors. By considering best relay selection beamforming criterion giving best performance in terms of OP considering relay, source and destination, analytical expression of OP was calculated and validated through simulations. In [130], authors developed a SWIPT based protocol by considering non-linear energy harvesting model in massive MIMO systems. Here a more realistic power consumption scheme was suggested. Particularly, after harvesting the energy, a part of energy harvested is employed for basic processing, e.g. consumption of energy for channel estimation, energy consumed by the hardware circuitry and feedback from the channel. The remaining energy is used for subsequent information transmission. Authors proposed a beam-domain SWIPT protocol for massive MIMO system and the SE of the system was maximized by finding optimized values of TS ratio and BS transmit power. The results showed that by employing Basis Expansion Model (BEM) great reduction in size of channel that requires estimation can be achieved together with number of CSIs needing feedback, thereby reducing feedback overhead for downlink estimation of channel.

F. SWIPT COR with hardware impairments (HI):

In order to physically implement RF transmitters we need to consider a number of impairments that deteriorate the quality of signal received, apart from the effects of fading and thermal noise at the receivers [131]. RF transmitters use Power Amplifiers (PAs) to guarantee that signals are communicated at the desired power levels. Due to this fact, PAs are operated with large input power resulting in undesired and nonlinear distortions. Although pre-distortion or calibration algorithms, can be utilized to partially mitigate of these problems and their impact, but still these Hardware Impairments (HIs) cannot be fully removed. Added impairments owing to low quality hardware e.g. I/Q imbalance, phase noise, sampling rate offset and carrier frequency as well as impedance mismatch together with other rapidly changing nonlinearities further degrade the performance of systems [132].

Recently, there has been a lot of research on performance analysis and other related issues in several cooperative relaying systems which consider the involvement of HIs. In [133], authors discussed the OP analysis of dual hop MIMO systems having AF relaying and involving HIs. By considering fixed as well as variable gain relaying beamforming algorithms, OP expressions were analyzed at SNR. Results showed the effects of HIs and number of antennas employed at relay on performance of system. Authors in [134], discussed the outage performance of a PS based two way AF SWIPT cooperative network with Time Division Broadcast (TDBC) and Nakagami-m fading channel, including HIs at the transceivers. An expression in closed form for system's OP based on HIs was derived. Results showed that optimal PS ratio increases when there is an increase in shape parameter of relay links and there is a decrease in OP as soon as the relay is located closer to that terminal which a lesser shape parameter of its relay link.

Apart from HIs, imperfect CSI (ICSI) is also a critical factor affecting the performance of various 5G and beyond wireless systems. NOMA based AF relay system in the presence of residual hardware impairments (RHIs), Channel Estimation Errors (CEE) and imperfect Successive Interference Cancellation (ipSIC) were discussed in [135]. By considering OP and EE as performance metrics, it is observed that RHIs, ipSIC and CEE all have an adverse effect on performance of the system. Additionally, it was depicted that CEEs have more effect on OP than RHIs. In [136], Li et.al discussed the effects of HIs and ICSI (imperfect channel state information) by considering a cooperative multi-relay system with NOMA and EH.OP was derived using PSR protocol. Simulation results showed that HIs have a considerable effect on OP performance and ICSI has a negative impact resulting in an increase in OP, the effects of which can be greatly reduced by an increase in number of relays. In [137], authors suggested a SWIPT based NOMA network where EH is done using TSR protocol by taking into account ICSI and residual hardware impairments (RHIs).OP as well as throughput in (delay limited transmission mode) were the performance metrics. It is depicted that performance of system is greatly limited by ICSI and RHIs resulting in an error floor for OP and diversity order gets zero. Additionally, it is seen that the throughput can be maximized by carefully determining the TS parameter.

G. Exploring SWIPT at 5G and B5G new frequencies:

5G and B5G is expected to offer end users with Gbps data rate as well as minimum latency, so that various new applications can be supported anytime anywhere. Nevertheless, the requirement of high data rate typically results in a greater consumption of energy, which reduces the QoS experience of users, specifically battery equipped powered devices. Thus some IoT sensors and wearable devices suffer from energy insufficiency as recharging is often very difficult in certain situations. Therefore, it has become a notable concern in optimization and design of B5G wireless communication as how to manage the conflict between conservation of energy and rate enhancement, where in SWIPT can play its vital role.

Although a lot of research has been made on integration of SWIPT with various multiple access technologies e.g. orthogonal frequency-division multiple access (OFDMA), time-division multiple access (TDMA) and NOMA. However most works in literature studied SWIPT only in the low frequency (LF) band. For satisfying the extremely high data rate requirements of 5G and B5G, high frequency (HF) band, i.e., mmWave, becomes unavoidable choice for enhancing available spectrum since LF band has a limited bandwidth. Furthermore, the channel propagation characteristics of some new frequencies that are considered in 5G and shall be exploited in B5G are entirely different from traditional LF channel models specifically lower than 3 GHz [138]. Thus it is imperative to determine correct channel propagation characteristics in order for SWIPT to work at new frequencies in both low and high frequency bands. In [139], authors examined SWIPT at 5G new frequency bands comprising measurement of channel parameters by considering large scale fading in both LF (3.5GHz) and HF (28GHz) spectrum, network design and optimization by taking into account both indoor and outdoor setups. A comparison of propagation characteristics of these two frequency bands was discussed and a more precise channel model was developed. For enhancement of EH efficiency of the proposed SWIPT network, a dual band network model was suggested with an optimized power allocation scheme and channel assignment for every user

IEEEAccess

VI. SWIPT CoR FOR INTERNET OF THINGS (IoT)

IoT is an evolving technique providing smart exchange of information for various intelligent devices in future networks including 5G and B5G [140]. With the advancements in IoT, energy as well as spectral efficiency are the two vital concerns owing to a widespread increase in the number of IoT devices. The aim of IoT is to establish a connection between physical and virtual environment to realize a self-sustaining system [141]. In order to attain this integration in future, many research works have been done to accomplish the application of IoT technology in smart cities, smart energy, smart homes, smart commutation and many more. The evolution of IoT originated from WSN and established into various heterogeneous networks capable of communicating with each other by utilizing Internet [142].While WSN networks are operating in their private networks, IoT accumulates numerous sensor networks in an IoT network which can access the Internet [143].



Fig. 21: SWIPT CoR for IoT

The rapidly developing IoT region is aiming to accelerate the consumption of energy required by small devices drastically. Since many sensors are being employed with power consumption of order of 1 W or less, the huge amount of such devices approx. 25 billion is aiming to utilize greater power as compared to cellular networks throughout the world [144]. Consequently, it is imperative to deal with this evolving challenge of EE in all of these small devices by employing an emerging RF harvesting technique called SWIPT. Apart from prolonging their lifetime, cooperative relaying has been employed for IoT systems with SWIPT, primarily to improve their coverage and capacity along with self-sustainability. Thus SWIPT CoR for next generation wireless networks particularly for IoT is an emerging area of research. Guo et al. in [144] discussed optimal scheduling of information and power transfer by the transmitter and introduced the concept of crowd harvesting, i.e., harvesting from other sources apart from transmitters. They discussed SWIPT and non-SWIPT based relaying where non-SWIPT case can make use of crowd harvesting from wireless access points, TV and BS downlinks. For Line Of Sight (LOS) TV bands are helpful sources of EH while for non-LOS Wi-Fi can play its part. SWIPT CoR for IoT is shown in Fig. 21.

Authors in [145] discussed resource management scheme in a DF cooperative IoT network. They proposed an optimization scheme to maximize overall sum rate considering proper relay assignment, power splitting ratio selection and power allocation. Results showed that CoR with diversity increases the data rates for individual IoT devices while enhancing the sum rate of overall network. Lu *et* al. devised an optimization problem for transmission rate maximization in a multi-relays Orthogonal Frequency Division Multiplexing (OFDM) based IoT network [146]. Simulation results based on effect of number of relays, conversion efficiency and power splitting ratio on transmission rate proved that the rate of transmission is increased considerably by the suggested scheme.

In [147], relay selection criterion based upon residual

energy at the relays for IoT system was suggested. Using stochastic geometry approach large and small scale fading effects were encountered. OP based on simple relay selection was derived based on approximate relay location. An Ultra-Reliable Low Latency Communication (ULLC) network was considered in [148] where reliability of IoT based CoR network was improved by optimized parameters for TS and PS along with finite block length codes.

In [149], an EH CoR system comprising of source, destination as well as multiple relays with DF protocol was considered by employing PSR protocol to harvest energy. A comparison between equal Power-Splitting-Based Relay Selection (EPS-RS) vs Optimal Power-Splitting-Based Relay Selection (OPS-RS) was made. They also proposed optimal relay selection for EH relays having batteries to store energy harvested. Results showed that battery enabled optimal power splitting relay selection outperformed the conventional equal power splitting based relay selection without batteries.

Authors in [150], examined a DF SWIPT with multi-hop relays considering IoT sensor network. A node with single antenna was considered with two optimization problems. The first problem optimized PS ratio by considering minimization of source transmit power and then as a second case the PS ratio was optimized with the objective of maximum achievable rate. Along with obtaining closed form solutions a few extension in their work have also been suggested.

A. SWIPT CoR with HI for B5G IoT:

One of the most challenging task for 5G and beyond 5G (B5G) is the connectivity of enormously large number of intelligent devices i.e IoT. The extensive applications of IoT and massive IoT in various fields, such as industrial applications, medicine, agriculture and many more have stimulated an extensive increase in number of IoT devices. Thus B5G wireless communication networks shall be facing large number of issues, such as energy consumption requirements and efficient spectrum utilization specifically for massive IoT. Recent studies involving IoT focused on ideal hardware circuitry and perfect CSI. However, nonlinearity in the hardware as well as imperfections in channel estimates must be accounted for in order to develop a close approximation of realistic environments. In [151], authors determined the performance of cooperative SWIPT IoT systems based on NOMA, in the presence of RHI and CEE. By employing PRS protocol, OP for Near User (NU) and Far User (FU) as well as diversity orders in high SNR region were studied. Results showed that CEE has an adverse effect on system's performance causing an increase in OP, but by increasing number of relays, effect of HI and CEE can be reduced.

Authors in [152] discussed a cooperative NOMA system for IoT with a master node serving multiple client nodes in the presence of HIs. The relay node can operate in both half as well as full duplex mode. The performance metrics are This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703. IEEE Access

IEEEAccess

OP and ergodic capacity, and results showed that there is a degradation in performance owing to HIs. The combined effect of in phase and quadrature phase imbalance (IQI) and ipSIC on NOMA based cooperative IoT network is discussed in [153]. Performance metrics are OP for near and far IoT, throughput and EE in the presence of HIs. It is depicted from simulation results that compared ipSIC has a greater effect on OP for the near IoT device as compared to IQI of the proposed system. Results show that OP of IoT devices can be decreased by designing a specific scheme for power allocation.

Authors in [154] investigated the performance of SWIPT with cooperative relaying and NOMA in the presence of CEE and HIs for IoT system. Multiple relays were employed and optimal relay was selected based on partial relay selection (PRS) protocol to forward the received signal to near and far users in proposed IoT network. Simulation results depicted that hardware impairment parameter (k) has a negative impact on system performance whereas the channel estimation parameter (δ) was found beneficial for OP. Security aspects exploiting the existence of eavesdroppers for B5G networks were discussed in [155]. A framework was developed to estimate the security and reliability of SWIPT based multi-relay network employing DF relays. Analysis was based on nonlinear energy harvesters, CEE and IQI. Analytical expressions for OP and intercept probability (IP) were obtained and simulation results depicted that IQI mismatch decreases OP performance and provides IP performance improvement. Also by increasing the number of relays better OP can be achieved.

Authors in [156] suggested SWIPT based EH by employing TSR, PSR and NOMA in the presence of interference for IoT cooperative relay systems. Novelty of their work was that they developed a model in which data was transmitted not only by source node but IoT relay node data was also forwarded by employing NOMA protocol to the destination. OP and throughput were the performance metrics. Authors suggested a very realistic scheme that could be simply employed for various IoT applications e.g underground tunnels, coal mines, under water applications as well as hazardous and toxic surroundings where IoT devices are utilized as detectors.

VII. FUTURE CHALLENGES

There are a few future challenges in SWIPT based relay networks which have to be explored in order to get better performance. The brief review of major challenges is outlined below:

1) Interference Management:

One of the significant challenges in SWIPT scheme is exploiting the interference, which involves proper interference management as it is supposed to be a harmful factor affecting the QoS of the information signal. However strong interference could be useful in increasing the energy harvested for an EH receiver. A challenging task is to utilize interference from various sources e.g in a large scale IoT network to its maximum for EH by keeping QoS constraints on ID [157].

2) Resource Allocation:

Resource allocation is a critical factor to enhance performance of SWIPT systems. Optimized resource scheduling mechanisms need to be developed based on maximum harvested energy and QoS constraint along with minimum transmission power of the source. Although a lot of research is ongoing on this area [158], [159] still new algorithms need to be developed specifically for massive IoT networks.

3) CSI Estimation:

When considering a wireless communication network taking into account effects of fading is of prime importance. If availability of CSI is made at the transmitter, the transmitter can adjust it's transmit power according to channel state such that there is no wastage of energy for transmission when deep channel fading is there. A more challenging task is CSI acquisition using training sequences and their management for large scale IoT network with billions of devices connected together and exchanging formation.

4) Relay Selection Optimization:

When considering a multi relay CoR network with SWIPT for EH, a major task is to optimally select relay for both EH and ID. As it is seen in literature [13], that the relay which is suitable for ID may not be optimal for harvesting energy. Although existing researches explore the idea of optimal relay selection in SWIPT systems, yet more algorithms needs to be developed so as to satisfy EH and ID constraints at the same time with proper relay selection specially when considering an IoT framework.

5) Security Issues:

In SWIPT with CoR, although the relay is a cooperative node, it may be a malicious node or an eavesdropper. Thus an untrusted relay may escalate probability of interception considerably. Use of massive MIMO techniques to improve information transmission security and to increase power transfer efficiency simultaneously have been discussed in [150]. Although lots of research work is done in this field, still more attention has to be given to develop adaptive and dynamic security algorithms for energy harvesting cooperative SWIPT networks.

6) Massive MIMO and massive access for B5G networks:

Multi antenna technology has certainly been a vital part of 5G networks, but it is equally important for researchers all over to globe to devise new multiple antenna schemes so as



to obtain the enormously large data rate, traffic loads and reliability in the B5G networks. For B5G, massive MIMO involving an array of at least 64 antennas has gained significant attention. In [160], authors reviewed multiple antenna technologies that play a massive role in B5G networks namely cell-free massive MIMO, beamspace massive MIMO, and IRS. Cell-free massive MIMO comprises of many APs with one or more antennas connected to a central unit through optical fiber links [161]. Cell-free massive MIMO widens the coverage area and has a high spectral efficiency. The extensive uses of IoT in various fields, such as agriculture, telemedicine as well as traffic have stimulated a drastic increase in number of IoT devices. In [162], authors discussed massive connectivity for 5G and B5G wireless networks. The massive IoT applications can acquire access to numerous wireless networks primarily through low-cost technologies e.g ZigBee, WiFi and Bluetooth. In order to have a broader coverage for IoT, presently, cellular IoT and Long Range Radio (LoRa) are two major access techniques. As compared to LoRa technology, cellular IoT is more helpful and cost-effective for service providers because it reutilizes prevailing cellular infrastructure. They have provided a detailed review on massive access in B5G cellular IoT involving less power, enormous connectivity as well as wider coverage. Massive access is still an emerging area of research in the context of SWIPT and CoR.

7) SWIPT CoR for mmWave:

Owing to enormous available bandwidth potential, the mmWave has a huge potential to fulfill the demand of the next generation wireless networks [163]. The mmWave communications suffer from major technical challenges like severe path loss, directivity issues, sensitivity to blockage and smaller beamwidth because of short wavelengths [164]. The mmWave and its integration with SWIPT CoR is relatively a new research area which has gained swelling interest by researchers in detail. In [165], authors discussed the characteristics of 5G mmWave systems and provided a general framework of SWIPT assisted mmWave network. . In [166], a rectenna at 24 GHz was successfully designed and the possibility of WPT towards mmWave technology was demonstrated. Also, a harvest-use strategy was suggested in [166], where harvesting energy from BS via mmWave signals and then consuming the harvested energy to transmit data was proposed. In [167], authors discussed a scheme by which a joint optimization of the transmit power, beamforming, and power splitting ratio for a lens-array antenna structure leading to maximization of energy efficiency of SWIPT systems was suggested . Kamga and Aissa [168] examined the performance of WPT in mmWave massive MIMO, working in both rainy as well as clear conditions. It was demonstrated that rain attenuation has a significant effect on EH efficiency in HF band and can also make WPT difficult in severe cases. The performance of relay assisted mmWave massive MIMO system was suggested in [169]. In a similar system, authors in [170] examined the EH capability of mmWave by considering the stochastic geometry approach. Optimized power allocation algorithms still have to be developed so that enhancement in performance of mmWave based SWIPT systems can be made with particular focus on a network of largely connected IoT devices. Future challenges in terms of SWIPT based mmWave communications include exploiting the diversity and spatial multiplexing gain for massive MIMO systems, power splitting and transceiver design, development of hybrid precoding model and exploring the design of RF hardware prototypes. In summary, extensive research needs to be carried out to explore the benefits of mmWave SWIPT systems involving CoR in future.

8) SWIPT for Underwater Internet of Things (UIoT):

An expansion of the IoT in the submarine surroundings, is called the underwater IoT, which has started to become a progressively more leading technology to develop the smart ocean environment [171]. Underwater IoT (UIoT) is a smart network having autonomously learning intelligent computing abilities. The sensing, monitoring, and identification of underwater objects can be done in UIoT by employing wired or wireless communication technologies. Hence, the model and design of the land-based IoT cannot be adopted directly by the UIoT. For instance, ocean currents constitute a non-negligible issue for the deployment of the UIoT [172]. Underwater node movement in the UIoT caused by ocean currents often affects network coverage and data transmission quality. In addition, the battery power of UIoT sensor nodes is severely limited, and node batteries cannot be conveniently recharged owing to the limitation of seawater corrosion and seawater pressure. A challenging task is to design a framework for recharging the batteries of underwater equipment [173]. Hence, underwater research and development could have a significant impact on many aspects of human's life by establishing and rolling out the UIoT. Therefore the design ,analysis and implementation of SWIPT protocols for energy efficient sensor nodes employed underwater is an open challenge for 5G and B5G networks.

VIII. CONCLUSION

This article reviewed the important concepts of SWIPT and its applications in wireless networks. SWIPT is an emerging technology that enables the transmission of energy along with the information at the same time providing spectral and energy efficient resolution considering 5G and beyond wireless networks. This article particularly focused on SWIPT, its architecture and its application in IoT networks. Moreover, it also provided a review on CoR and its use in IoT technology. The combination of SWIPT with CoR for IoT networks is also presented in detail. Although a large This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703, IEEE Access

IEEE Access Multidisciplinary : Rapid Review : Open Access Journal

number of opportunities as well challenges are coming forward in SWIPT based CoR, some of them have been stated above. However, greater attention and investigations are required to completely explore the two conspicuous technologies for additional development of future generations of wireless networks.

APPENDIX

	TABLE II
A LIST	OF ACRONYMS USED IN THIS MANUSCRIPT.

Acronyms	Definition			
4 G	Fourth Generation			
5G	Fifth Generation			
AF	Amplify and Forward			
AS	Antenna Switching			
AWGN	Additive White Gaussian Noise			
B5G	Beyond Fifth Generation			
BEM	Basis Expansion Model			
BER	Bit Error Rate			
BS	Base Station			
CF	Code & Forward relaying protocol			
CoR	Cooperative Relaying			
CSI	Channel State Information			
CW	Continuous Wave			
D2D	Device to Device			
DA	Distributed Antenna			
DAS	Distributed Antenna System			
DC	Direct Current			
DF	Decode and Forward			
DL	Direct Link			
EE	Electrical Efficiency			
EH	Wireless Energy Harvesting			
EPS-RS	Equal Power-Splitting-Based Relay Selection			
FAF	Fixed Amplify & Forward Relaying Protocol			
FDF	Fixed Decode &Forward Relaying Protocol			
GRS	Global Relay Selection			
HDAF	Hybrid decode-amplify-and-forward			
HI	Hardware Impairments			
ІоТ	Internet of Things			
IR	Incremental relaying protocol			
IRS	Intelligent Reconfigurable Surface			
LES	Low Earth Satellite			
LOS	Line Of Sight			
M2M	Machine-to-Machine			

MIMO	Multiple-Input-Multiple-Output			
MIMO-BC	Multiple-Input Multiple-Output			
мию-вс	Broadcast Channel			
mmWave	Millimeter-Wave			
MRC	Maximum Ratio Combining			
MRT	Maximum Ratio Transmit			
NB-IoT	Narrowband IoT			
NOMA	Non-Orthogonal Multiple Access			
OFDM	Orthogonal Frequency Division			
	Multiplexing			
OP	Outage Probability			
OPS-RS	Optimal Power-Splitting-Based Relay			
D 4	Selection			
PA DC	Power Amplifier			
PS DCD	Power Splitting			
PSR	Power Splitting Relaying			
QoS	Quality-of-Service			
R-E	Rate-Energy			
RF	Radio Frequency			
RF-EH	Radio Frequency Energy Harvesting			
RNs	Relay Nodes			
RS	Relay Selection			
SDF	Selective Decode & Forward Relaying Protocol			
SE	Spectral Efficiency			
SINR	Signal to Interference and Noise Ratio			
SISO	Single-Input-Single-Output			
SIMO	Single-Input-Multiple-Output			
SNR	Signal to Noise Ratio			
SRS	Single Relay Selection			
SWIDT	Simultaneous Wireless Information and			
Swif I	Power Transfer			
TDBC	Time Division Broadcast			
TS	Time Switching			
TSR	Time Switching Relaying			
TSR-PSR	Hybrid Protocol			
UIoT	Underwater Internet of Things			
ULLC	Ultra-Reliable Low Latency			
ULLC	Communication			
WIT	Wireless Information Transfer			
WPCN	Wireless Powered Communication			
	Wireless Dower Transfor			
	Wireless Sonsor Networks			
<u>vv 51</u> ~	Time Switching Decemptor			
a	Time Switching Parameter			
ρ	Power Splitting Ratio			

This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703. IEEE Access

IEEE Access Multidisciplinary : Rapid Review : Open Access Journal

REFERENCES

- S. Atapattu and J. Evans, "Optimal power-splitting ratio for wireless energy harvesting in relay networks," in 2015 IEEE 82nd Vehicular Technology Conference, VTC Fall 2015 -Proceedings, Jan. 2016, doi: 10.1109/VTCFall.2015.7391056.
- [2] R. Q. Hu and Y. Qian, "An energy efficient and spectrum efficient wireless heterogeneous network framework for 5G systems," *IEEE Commun. Mag.*, vol. 52, no. 5, pp. 94–101, 2014, doi: 10.1109/MCOM.2014.6815898.
- [3] A. N. Kasim, I. Shayea, S. A. Khan, A. Alhammadi, and M. Ergen, "Evolutionary paths towards mobility management in 5G Heterogeneous Networks," in *Proceedings of 2020 IEEE Workshop* on Microwave Theory and Techniques in Wireless Communications, MTTW 2020, Oct. 2020, pp. 24– 29, doi: 10.1109/MTTW51045.2020.9245046.
- [4] X. Lu, P. Wang, D. Niyato, D. I. Kim, and Z. Han, "Wireless networks with rf energy harvesting: A contemporary survey," *IEEE Commun. Surv. Tutorials*, vol. 17, no. 2, pp. 757–789, 2015, doi: 10.1109/COMST.2014.2368999.
- [5] D. T. Hoang, D. Niyato, P. Wang, and D. I. Kim, "Opportunistic channel access and RF energy harvesting in cognitive radio networks," *IEEE J. Sel. Areas Commun.*, vol. 32, no. 11, pp. 2039– 2052, 2014, doi: 10.1109/JSAC.2014.141108.
- [6] P. Kamalinejad, C. Mahapatra, Z. Sheng, S. Mirabbasi, V. C. Victor, and Y. L. Guan, "Wireless energy harvesting for the Internet of Things," *IEEE Commun. Mag.*, vol. 53, no. 6, pp. 102–108, Jun. 2015, doi: 10.1109/MCOM.2015.7120024.
- [7] A. A. Nasir, X. Zhou, S. Durrani, and R. A. Kennedy, "Wireless-powered relays in cooperative communications: Time-switching relaying protocols and throughput analysis," *IEEE Trans. Commun.*, vol. 63, no. 5, pp. 1607–1622, 2015, doi: 10.1109/TCOMM.2015.2415480.
- [8] B. Clerckx, R. Zhang, R. Schober, D. W. K. Ng, D. I. Kim, and H. V. Poor, "Fundamentals of wireless information and power transfer: From RF energy harvester models to signal and system designs," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 1, pp. 4–33, Jan. 2019, doi: 10.1109/JSAC.2018.2872615.
- [9] A. M. Fouladgar and O. Simeone, "On the transfer of information and energy in multi-user systems," *IEEE Commun. Lett.*, vol. 16, no. 11, pp. 1733– 1736, 2012, doi: 10.1109/LCOMM.2012.091212.121660.
- I. Krikidis, S. Sasaki, S. Timotheou, and Z. Ding, "A low complexity antenna switching for joint wireless information and energy transfer in MIMO relay channels," *IEEE Trans. Commun.*, vol. 62, no. 5, pp. 1577–1587, 2014, doi: 10.1109/TCOMM.2014.032914.130722.

- Z. Xiang and M. Tao, "Robust beamforming for wireless information and power transmission," *IEEE Wirel. Commun. Lett.*, vol. 1, no. 4, pp. 372– 375, 2012, doi: 10.1109/WCL.2012.053112.120212.
- [12] R. Zhang and C. K. Ho, "MIMO broadcasting for simultaneous wireless information and power transfer," *IEEE Trans. Wirel. Commun.*, vol. 12, no. 5, pp. 1989–2001, 2013, doi: 10.1109/TWC.2013.031813.120224.
- [13] L. Liu, R. Zhang, and K. C. Chua, "Wireless information transfer with opportunistic energy harvesting," *IEEE Trans. Wirel. Commun.*, vol. 12, no. 1, pp. 288–300, 2013, doi: 10.1109/TWC.2012.113012.120500.
- [14] I. Krikidis, S. Timotheou, S. Nikolaou, G. Zheng, D. W. K. Ng, and R. Schober, "Simultaneous Wireless Information and Power Transfer in modern communication systems," *IEEE Commun. Mag.*, vol. 52, no. 11, pp. 104–110, Jan. 2014, doi: 10.1109/MCOM.2014.6957150.
- [15] I. Krikidis, "Simultaneous information and energy transfer in large-scale networks with/without relaying," *IEEE Trans. Commun.*, vol. 62, no. 3, pp. 900–912, 2014, doi: 10.1109/TCOMM.2014.020914.130825.
- [16] S. Bi, Y. Zeng, and R. Zhang, "Wireless powered communication networks: An overview," *IEEE Wirel. Commun.*, vol. 23, no. 2, pp. 10–18, 2016, doi: 10.1109/MWC.2016.7462480.
- J. N. Laneman, D. N. C. Tse, and G. W. Wornell, "Cooperative diversity in wireless networks: Efficient protocols and outage behavior," *IEEE Trans. Inf. Theory*, vol. 50, no. 12, pp. 3062–3080, Dec. 2004, doi: 10.1109/TIT.2004.838089.
- [18] S. Gautam and P. Ubaidulla, "Relay Selection and Transceiver Design for Joint Wireless Information and Energy Transfer in Cooperative Networks," in *IEEE Vehicular Technology Conference*, 2017, vol. 2017-June, pp. 1–5, doi: 10.1109/VTCSpring.2017.8108246.
- [19] C. Huang, R. Zhang, and S. Cui, "Throughput maximization for the gaussian relay channel with energy harvesting constraints," *IEEE J. Sel. Areas Commun.*, vol. 31, no. 8, pp. 1469–1479, 2013, doi: 10.1109/JSAC.2013.130811.
- [20] M. A. Hossain, R. M. Noor, K. L. A. Yau, I. Ahmedy, and S. S. Anjum, "A Survey on Simultaneous Wireless Information and Power Transfer with Cooperative Relay and Future Challenges," *IEEE Access*, vol. 7. Institute of Electrical and Electronics Engineers Inc., pp. 19166–19198, 2019, doi: 10.1109/ACCESS.2019.2895645.
- [21] D. N. K. Jayakody, J. Thompson, S. Chatzinotas, and S. Durrani, Wireless information and power transfer: A new paradigm for green

communications. Springer International Publishing, 2017.

- [22] H. S. Nguyen, T. S. Nguyen, and M. Voznak, "Relay selection for SWIPT: Performance analysis of optimization problems and the trade-off between ergodic capacity and energy harvesting," *AEU - Int. J. Electron. Commun.*, vol. 85, no. 1, pp. 59–67, 2018, doi: 10.1016/j.aeue.2017.12.012.
- [23] S. Bi, C. K. Ho, and R. Zhang, "Wireless powered communication: Opportunities and challenges," *IEEE Commun. Mag.*, vol. 53, no. 4, pp. 117–125, 2015, doi: 10.1109/MCOM.2015.7081084.
- [24] H. Lee, K. J. Lee, H. Kim, and I. Lee, "Wireless information and power exchange for energy-constrained device-to-device communications," *IEEE Internet Things J.*, vol. 5, no. 4, pp. 3175–3185, Aug. 2018, doi: 10.1109/JIOT.2018.2836325.
- [25] C. Zhang, W. Ahn, Y. Zhang, and B. R. Childers, "Live code update for IoT devices in energy harvesting environments," in 2016 5th Non-Volatile Memory Systems and Applications Symposium, NVMSA 2016, 2016, doi: 10.1109/NVMSA.2016.7547182.
- [26] N. Zhao, S. Zhang, F. R. Yu, Y. Chen, A. Nallanathan, and V. C. M. Leung, "Exploiting Interference for Energy Harvesting: A Survey, Research Issues, and Challenges," *IEEE Access*, vol. 5, pp. 10403–10421, 2017, doi: 10.1109/ACCESS.2017.2705638.
- [27] Z. Ding, I. Krikidis, B. Sharif, and H. V. Poor, "Wireless information and power transfer in cooperative networks with spatially random relays," *IEEE Trans. Wirel. Commun.*, vol. 13, no. 8, pp. 4440–4453, 2014, doi: 10.1109/TWC.2014.2314114.
- [28] T. D. P. Perera, D. N. K. Jayakody, S. De, and M. A. Ivanov, "A Survey on Simultaneous Wireless Information and Power Transfer," in *Journal of Physics: Conference Series*, 2017, vol. 803, no. 1, doi: 10.1088/1742-6596/803/1/012113.
- T. D. Ponnimbaduge Perera, D. N. K. Jayakody, S. K. Sharma, S. Chatzinotas, and J. Li, "Simultaneous Wireless Information and Power Transfer (SWIPT): Recent Advances and Future Challenges," *IEEE Commun. Surv. Tutorials*, vol. 20, no. 1, pp. 264–302, 2018, doi: 10.1109/COMST.2017.2783901.
- J. Huang, C. C. Xing, and C. Wang, "Simultaneous Wireless Information and Power Transfer: Technologies, Applications, and Research Challenges," *IEEE Commun. Mag.*, vol. 55, no. 11, pp. 26–32, 2017, doi: 10.1109/MCOM.2017.1600806.
- [31] R. Zhang, R. G. Maunder, and L. Hanzo, "Wireless information and power transfer: From scientific hypothesis to engineering practice," *IEEE Commun. Mag.*, vol. 53, no. 8, pp. 99–105, 2015,

doi: 10.1109/MCOM.2015.7180515.

- [32] Z. Ding *et al.*, "Application of smart antenna technologies in simultaneous wireless information and power transfer," *IEEE Commun. Mag.*, vol. 53, no. 4, pp. 86–93, 2015, doi: 10.1109/MCOM.2015.7081080.
- [33] M. L. Ku, W. Li, Y. Chen, and K. J. Ray Liu, "Advances in Energy Harvesting Communications: Past, Present, and Future Challenges," *IEEE Commun. Surv. Tutorials*, vol. 18, no. 2, pp. 1384–1412, 2016, doi: 10.1109/COMST.2015.2497324.
- [34] Y. Bhute, A. R. Raut, and A. Prof, "A Survey on Relay Selection Strategies in Cooperative Wireless Network for Capacity Enhancement," 2013.
- [35] Umamaheswari S and Sathya M, "A Comprehensive Survey on Cooperative Relaying in Industrial Wireless Sensor Network," *Int. J. Eng. Res.*, vol. V6, no. 07, pp. 591–595, 2017, doi: 10.17577/ijertv6is070285.
- [36] F. Mansourkiaie and M. H. Ahmed, "Cooperative routing in wireless networks: A comprehensive survey," *IEEE Commun. Surv. Tutorials*, vol. 17, no. 2, pp. 604–626, 2015, doi: 10.1109/COMST.2014.2386799.
- [37] C. X. Wang *et al.*, "Cellular architecture and key technologies for 5G wireless communication networks," *IEEE Commun. Mag.*, vol. 52, no. 2, pp. 122–130, 2014, doi: 10.1109/MCOM.2014.6736752.
- [38] K. Huang, C. Zhong, and G. Zhu, "Some new research trends in wirelessly powered communications," IEEE Wirel. Commun., vol. 23, 2. 19–27, Apr. 2016. doi: no. pp. 10.1109/MWC.2016.7462481.
- [39] L. R. Varshney, "Transporting information and energy simultaneously," in *IEEE International Symposium on Information Theory - Proceedings*, 2008, pp. 1612–1616, doi: 10.1109/ISIT.2008.4595260.
- [40] P. Grover and A. Sahai, "Shannon meets tesla: Wireless information and power transfer," in *IEEE International Symposium on Information Theory - Proceedings*, 2010, pp. 2363–2367, doi: 10.1109/ISIT.2010.5513714.
- [41] F. Mansourkiaie and M. H. Ahmed, "Cooperative routing in wireless networks: A comprehensive survey," *IEEE Commun. Surv. Tutorials*, vol. 17, no. 2, pp. 604–626, 2015, doi: 10.1109/COMST.2014.2386799.
- [42] U. Raza and A. Salam, "On-site and external energy harvesting in underground wireless," *Electronics* (*Switzerland*), vol. 9, no. 4. p. 681, 2020, doi: 10.3390/electronics9040681.
- [43] X. Zhou, R. Zhang, and C. K. Ho, "Wireless information and power transfer: Architecture design and rate-energy tradeoff," *IEEE Trans. Commun.*, vol. 61, no. 11, pp. 4754–4767, Nov. 2013, doi:

This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703. IEEE Access

IEEEAccess Multidisciplinary : Rapid Review : Open Access Journal

10.1109/TCOMM.2013.13.120855.

- [44] E. Boshkovska, D. W. K. Ng, N. Zlatanov, and R. Schober, "Practical non-linear energy harvesting model and resource allocation for SWIPT systems," *IEEE Commun. Lett.*, vol. 19, no. 12, pp. 2082– 2085, 2015, doi: 10.1109/LCOMM.2015.2478460.
- [45] Q. Shi, L. Liu, W. Xu, and R. Zhang, "Joint transmit beamforming and receive power splitting for MISO SWIPT systems," *IEEE Trans. Wirel. Commun.*, vol. 13, no. 6, pp. 3269–3280, 2014, doi: 10.1109/TWC.2014.041714.131688.
- [46] D. W. K. Ng, E. S. Lo, and R. Schober, "Wireless information and power transfer: Energy efficiency optimization in OFDMA systems," *IEEE Trans. Wirel. Commun.*, vol. 12, no. 12, pp. 6352–6370, 2013, doi: 10.1109/TWC.2013.103113.130470.
- [47] K. Xiong, B. Wang, and K. J. R. Liu, "Rate-Energy Region of SWIPT for MIMO Broadcasting under Nonlinear Energy Harvesting Model," *IEEE Trans. Wirel. Commun.*, vol. 16, no. 8, pp. 5147–5161, 2017, doi: 10.1109/TWC.2017.2706277.
- [48] E. Boshkovska, A. Koelpin, D. W. K. Ng, N. Zlatanov, and R. Schober, "Robust beamforming for SWIPT systems with non-linear energy harvesting model," in *IEEE Workshop on Signal Processing Advances in Wireless Communications, SPAWC*, Aug. 2016, vol. 2016-Augus, doi: 10.1109/SPAWC.2016.7536860.
- [49] Q. Shi, W. Xu, T. H. Chang, Y. Wang, and E. Song, "Joint beamforming and power splitting for MISO interference channel with SWIPT: An SOCP relaxation and decentralized algorithm," *IEEE Trans. Signal Process.*, vol. 62, no. 23, pp. 6194– 6208, 2014, doi: 10.1109/TSP.2014.2362092.
- [50] S. H. Cha, C. Jeong, and S. H. Lim, "Simultaneous wireless information and power transfer for internet of things sensor networks," *IEEE Internet Things J.*, vol. 5, no. 4, pp. 2829–2843, 2018, doi: 10.1109/JIOT.2018.2825334.
- [51] S. Timotheou, I. Krikidis, G. Zheng, and B. Ottersten, "Beamforming for MISO interference channels with QoS and RF energy transfer," *IEEE Trans. Wirel. Commun.*, vol. 13, no. 5, pp. 2646– 2658, 2014, doi: 10.1109/TWC.2014.032514.131199.
- [52] J. Tang, A. Shojaeifard, D. K. C. So, K. K. Wong, and N. Zhao, "Energy efficiency optimization with SWIPT in MIMO broadcast channels for internet of things," in *IEEE Internet of Things Journal*, 2018, vol. 5, no. 4, pp. 2605–2619, doi: 10.1109/JIOT.2017.2785861.
- [53] H. S. Lee and J. W. Lee, "Resource and task scheduling for SWIPT IoT systems with renewable energy sources," *IEEE Internet Things J.*, vol. 6, no. 2, pp. 2729–2748, 2019, doi: 10.1109/JIOT.2018.2873658.
- [54] D. S. Gurjar, H. H. Nguyen, and H. D. Tuan,

"Wireless information and power transfer for IoT applications in overlay cognitive radio networks," *IEEE Internet Things J.*, vol. 6, no. 2, pp. 3257–3270, Apr. 2019, doi: 10.1109/JIOT.2018.2882207.

- [55] M. Asshad, S. A. Khan, A. Kavak, and K. Kucuk, "Performance Analysis of Multi-Node Cooperative Network with Amplify and Forward Relay Protocol," in 1st International Informatics and Software Engineering Conference: Innovative Technologies for Digital Transformation, IISEC 2019 - Proceedings, Nov. 2019, doi: 10.1109/UBMYK48245.2019.8965574.
- [56] K. W. Choi *et al.*, "Simultaneous Wireless Information and Power Transfer (SWIPT) for Internet of Things: Novel Receiver Design and Experimental Validation," *IEEE Internet Things J.*, vol. 7, no. 4, pp. 2996–3012, 2020, doi: 10.1109/JIOT.2020.2964302.
- [57] Y. Huang, M. Liu, and Y. Liu, "Energy-efficient SWIPT in IoT distributed antenna systems," *IEEE Internet Things J.*, vol. 5, no. 4, pp. 2646–2656, Aug. 2018, doi: 10.1109/JIOT.2018.2796124.
- [58] J. M. Kang, C. J. Chun, I. M. Kim, and D. I. Kim, "Dynamic power splitting for SWIPT with nonlinear energy harvesting in ergodic fading channel," *IEEE Internet Things J.*, vol. 7, no. 6, pp. 5648–5665, Jun. 2020, doi: 10.1109/JIOT.2020.2980328.
- [59] H. Yang, Y. Ye, X. Chu, and M. Dong, "Resource and Power Allocation in SWIPT-Enabled Deviceto-Device Communications Based on a Nonlinear Energy Harvesting Model," *IEEE Internet Things J.*, vol. 7, no. 11, pp. 10813–10825, Nov. 2020, doi: 10.1109/JIOT.2020.2988512.
- [60] E. Boshkovska, N. Zlatanov, L. Dai, D. W. K. Ng, and R. Schober, "Secure SWIPT Networks Based on a Non-Linear Energy Harvesting Model," in 2017 IEEE Wireless Communications and Networking Conference Workshops, WCNCW 2017, May 2017, doi: 10.1109/WCNCW.2017.7919062.
- [61] A. Nosratinia, T. E. Hunter, and A. Hedayat, "Cooperative communication in wireless networks," *IEEE Commun. Mag.*, vol. 42, no. 10, pp. 74–80, Oct. 2004, doi: 10.1109/MCOM.2004.1341264.
- [62] T. M. Cover and A. A. El Gamal, "Capacity Theorems for the Relay Channel," *IEEE Trans. Inf. Theory*, vol. 25, no. 5, pp. 572–584, 1979, doi: 10.1109/TIT.1979.1056084.
- [63] J. N. Laneman, "Cooperative Diversity in Wireless Networks: Algorithms and Architectures," Massachusetts Institute of Technology, 1995. Accessed: Apr. 06, 2021. [Online]. Available: https://dspace.mit.edu/handle/1721.1/29272.
- [64] Q. Li, R. Q. Hu, Y. Qian, and G. Wu, "Cooperative communications for wireless networks: Techniques

and applications in LTE-advanced systems," *IEEE Wireless Communications*, vol. 19, no. 2. pp. 22–29, 2012, doi: 10.1109/MWC.2012.6189409.

- [65] L. C. Tran, A. Mertins, X. Huang, and F. Safaei, "Comprehensive performance analysis of fully cooperative communication in WBANs," *IEEE Access*, vol. 4, pp. 8737–8756, 2016, doi: 10.1109/ACCESS.2016.2637568.
- [66] E. Zimmermann, P. Herhold, and G. Fettweis, "On the performance of cooperative relaying protocols in wireless networks," in *European Transactions on Telecommunications*, Jan. 2005, vol. 16, no. 1, pp. 5–16, doi: 10.1002/ett.1028.
- [67] J. Nicholas Laneman and G. W. Wornell, "Energyefficient antenna sharing and relaying for wireless networks," in 2000 IEEE Wireless Communications and Networking Conference, 2000, pp. 7–12, doi: 10.1109/wcnc.2000.904590.
- [68] Y. Jing and H. Jafarkhani, "Single and multiple relay selection schemes and their achievable diversity orders," *IEEE Trans. Wirel. Commun.*, vol. 8, no. 3, pp. 1414–1423, 2009, doi: 10.1109/TWC.2008.080109.
- [69] H. Chen, L. Xiao, D. Yang, T. Zhang, and L. Cuthbert, "User Cooperation in Wireless Powered Communication Networks with a Pricing Mechanism," *IEEE Access*, vol. 5, pp. 16895– 16903, 2017, doi: 10.1109/ACCESS.2017.2740403.
- [70] S. Efazati and P. Azmi, "Cross Layer Power Allocation for Selection Relaying and Incremental Relaying Protocols over Single Relay Networks," *IEEE Trans. Wirel. Commun.*, vol. 15, no. 7, pp. 4598–4610, 2016, doi: 10.1109/TWC.2016.2542814.
- [71] S. Agnihotri, S. Jaggi, and M. Chen, "Amplify-andforward in wireless relay networks," in 2011 IEEE Information Theory Workshop, ITW 2011, 2011, pp. 311–315, doi: 10.1109/ITW.2011.6089444.
- [72] Vishal K. Shah and Anuradha P. Gharge, "A Review on Relay Selection Techniques in Cooperative," *Int. J. Eng. Innov. Technol.*, vol. 2, no. 5, pp. 65–69, 2012.
- [73] K. J. R. Liu, A. K. Sadek, W. Su, and A. Kwasinski, *Cooperative communications and networking*, vol. 9780521895. Cambridge University Press, 2008.
- [74] R. Shankar, G. Kumar, V. Sachan, and R. K. Mishra, "An Investigation of Two Phase Multi-Relay S-DF Cooperative Wireless Network over Time-Variant Fading Channels with Incorrect CSI," in *Procedia Computer Science*, 2018, vol. 125, pp. 871–879, doi: 10.1016/j.procs.2017.12.111.
- [75] S. S. Ikki and M. H. Ahmed, "Performance analysis of incremental-relaying cooperative-diversity networks over Rayleigh fading channels," *IET Commun.*, vol. 5, no. 3, pp. 337–349, Feb. 2011, doi: 10.1049/iet-com.2010.0157.

- [76] Q. Wu and R. Zhang, "Intelligent Reflecting Surface Enhanced Wireless Network: Joint Active and Passive Beamforming Design," in 2018 IEEE Global Communications Conference, GLOBECOM 2018 - Proceedings, 2018, doi: 10.1109/GLOCOM.2018.8647620.
- [77] X. Yu, D. Xu, Y. Sun, D. W. K. Ng, and R. Schober, "Robust and Secure Wireless Communications via Intelligent Reflecting Surfaces," *IEEE J. Sel. Areas Commun.*, vol. 38, no. 11, pp. 2637–2652, Nov. 2020, doi: 10.1109/JSAC.2020.3007043.
- [78] E. Basar, M. Di Renzo, J. De Rosny, M. Debbah, M. S. Alouini, and R. Zhang, "Wireless communications through reconfigurable intelligent surfaces," *IEEE Access*, vol. 7, pp. 116753–116773, 2019, doi: 10.1109/ACCESS.2019.2935192.
- [79] Y. Tang, G. Ma, H. Xie, J. Xu, and X. Han, "Joint Transmit and Reflective Beamforming Design for IRS-Assisted Multiuser MISO SWIPT Systems," in *IEEE International Conference on Communications*, Jun. 2020, vol. 2020-June, doi: 10.1109/ICC40277.2020.9148892.
- [80] C. Huang, A. Zappone, G. C. Alexandropoulos, M. Debbah, and C. Yuen, "Reconfigurable intelligent surfaces for energy efficiency in wireless communication," in *IEEE Transactions on Wireless Communications*, Aug. 2019, vol. 18, no. 8, pp. 4157–4170, doi: 10.1109/TWC.2019.2922609.
- [81] Q. Wu and R. Zhang, "Weighted Sum Power Maximization for Intelligent Reflecting Surface Aided SWIPT," *IEEE Wirel. Commun. Lett.*, vol. 9, no. 5, pp. 586–590, May 2020, doi: 10.1109/LWC.2019.2961656.
- [82] G. Yu, X. Chen, C. Zhong, D. W. K. Ng, and Z. Zhang, "Design, Analysis, and Optimization of a Large Intelligent Reflecting Surface-Aided B5G Cellular Internet of Things," *IEEE Internet Things J.*, vol. 7, no. 9, pp. 8902–8916, Sep. 2020, doi: 10.1109/JIOT.2020.2996984.
- [83] S. Sonmez, I. Shayea, S. A. Khan, and A. Alhammadi, "Handover Management for Next-Generation Wireless Networks: A Brief Overview," in *Proceedings of 2020 IEEE Workshop on Microwave Theory and Techniques in Wireless Communications, MTTW 2020*, Oct. 2020, pp. 35–40, doi: 10.1109/MTTW51045.2020.9245065.
- [84] S. A. Chaudhry, G. Üniversitesi, M. H. Alsharif, and K. Yahya, "Analyzing and evaluating the energy efficiency based on multi-5G small cells with a mm-waves in the next generation cellular networks," *Artic. Int. J. Electr. Comput. Eng.*, vol. 10, no. 4, pp. 3492–3500, 2020, doi: 10.11591/ijece.v10i4.pp3492-3500.
- [85] N. C. Beaulieu and J. Hu, "A closed-form expression for the outage probability of decodeand-forward relaying in dissimilar Rayleigh fading channels," *IEEE Commun. Lett.*, vol. 10, no. 12, pp.

813–815, 2006, doi: 10.1109/LCOMM.2006.061048.

- [86] S. S. Ikki and M. H. Ahmed, "Performance analysis of multi-branch decode-and-forward cooperative diversity networks over Nakagami-m fading channels," in *IEEE International Conference on Communications*, 2009, doi: 10.1109/ICC.2009.5199311.
- [87] A. Dziri, M. Terre, and N. Nasser, "Performance analysis of relay selection for IoT networks over generalized k distribution," in 2019 15th International Wireless Communications and Mobile Computing Conference, IWCMC 2019, 2019, pp. 1411–1415, doi: 10.1109/IWCMC.2019.8766438.
- [88] G. Yang, T. Liang, X. He, Y. Song, and C. Wu, "Relay-based Cooperative Communication Framework for Narrowband Internet of Things," in 2019 IEEE International Conference on Consumer Electronics - Taiwan, ICCE-TW 2019, May 2019, doi: 10.1109/ICCE-TW46550.2019.8991952.
- [89] A. Alsharoa, X. Zhang, D. Qiao, and A. Kamal, "An energy-efficient relaying scheme for internet of things communications," in *IEEE International Conference on Communications*, 2018, vol. 2018-May, doi: 10.1109/ICC.2018.8422169.
- [90] J. Bao, J. Wu, C. Liu, B. Jiang, and X. Tang, "Optimized Power Allocation and Relay Location Selection in Cooperative Relay Networks," *Wirel. Commun. Mob. Comput.*, vol. 2017, pp. 1–10, 2017, doi: 10.1155/2017/9727360.
- [91] S. A. Chaudhry, K. Yahya, F. Al-Turjman, and M. H. Yang, "A Secure and Reliable Device Access Control Scheme for IoT Based Sensor Cloud Systems," *IEEE Access*, vol. 8, pp. 139244–139254, 2020, doi: 10.1109/ACCESS.2020.3012121.
- [92] C. Zhong, H. A. Suraweera, G. Zheng, I. Krikidis, and Z. Zhang, "Wireless information and power transfer with full duplex relaying," *IEEE Trans. Commun.*, vol. 62, no. 10, pp. 3447–3461, 2014, doi: 10.1109/TCOMM.2014.2357423.
- [93] A. A. Nasir, X. Zhou, S. Durrani, and R. A. Kennedy, "Relaying protocols for wireless energy harvesting and information processing," *IEEE Trans. Wirel. Commun.*, vol. 12, no. 7, pp. 3622–3636, 2013, doi: 10.1109/TWC.2013.062413.122042.
- [94] S. Atapattu, H. Jiang, J. Evans, and C. Tellambura, "Time-switching energy harvesting in relay networks," in *IEEE International Conference on Communications*, 2015, vol. 2015-Septe, pp. 5416– 5421, doi: 10.1109/ICC.2015.7249185.
- [95] S. Atapattu and J. Evans, "Optimal Energy Harvesting Protocols for Wireless Relay Networks," *IEEE Trans. Wirel. Commun.*, vol. 15, no. 8, p. 5789, 2016, doi: 10.1109/TWC.2016.2569097.
- [96] Y. Liu, R. Xiao, J. Shen, H. Yang, and C. Yan,

"Hybrid protocol for wireless energy harvesting network over log-normal fading channel," *J. Eng.*, vol. 2018, no. 6, pp. 339–341, 2018, doi: 10.1049/joe.2017.0892.

- [97] A. A. Nasir, X. Zhou, S. Durrani, and R. A. Kennedy, "Throughput and ergodic capacity of wireless energy harvesting based DF relaying network," in 2014 IEEE International Conference on Communications, ICC 2014, 2014, pp. 4066–4071, doi: 10.1109/ICC.2014.6883957.
- [98] T. Wang, "Comparison of the energy efficiency for decode-and-forward and amplify-and-forward twoway relaying," in *Proceedings of 2013 5th IEEE International Conference on Broadband Network and Multimedia Technology, IEEE IC-BNMT 2013*, 2013, pp. 232–236, doi: 10.1109/ICBNMT.2013.6823948.
- [99] Y. Gu and S. Aïssa, "RF-Based Energy Harvesting in Decode-and-Forward Relaying Systems: Ergodic and Outage Capacities," *IEEE Trans. Wirel. Commun.*, vol. 14, no. 11, pp. 6425–6434, Nov. 2015, doi: 10.1109/TWC.2015.2453418.
- [100] F. Wang, W. Xu, S. Li, Z. Feng, and J. Lin, "Outage probability analysis of DF relay networks with RF energy harvesting," in 2015 IEEE Global Communications Conference, GLOBECOM 2015, 2015, pp. 1–5, doi: 10.1109/GLOCOM.2014.7417579.
- [101] S. T. Shah, D. Munir, M. Y. Chung, and K. W. Choi, "Information processing and wireless energy harvesting in two-way amplify-and-forward relay networks," in *IEEE Vehicular Technology Conference*, Jul. 2016, vol. 2016-July, pp. 1–5, doi: 10.1109/VTCSpring.2016.7504290.
- [102] A. Zahedi, M. Ergen, and I. Shayea, "Optimum time/power fraction of energy harvesting in TSR/PSR SWIPT-based cooperative communications with effective capacity maximization approach," AEU - Int. J. Electron. Commun.. vol. 111. 2019. doi: 10.1016/j.aeue.2019.152889.
- [103] T. Yuan, M. Liu, and Y. Feng, "Performance Analysis for SWIPT Cooperative DF Communication Systems with Hybrid Receiver and Non-Linear Energy Harvesting Model," *Sensors*, vol. 20, no. 9, p. 2472, Apr. 2020, doi: 10.3390/s20092472.
- [104] G. Pan and C. Tang, "Outage performance on threshold AF and DF relaying schemes in simultaneous wireless information and power transfer systems," *AEU - Int. J. Electron. Commun.*, vol. 71, pp. 175–180, 2017, doi: 10.1016/j.aeue.2016.10.021.
- [105] S. Mahama, D. K. P. Asiedu, and K. J. Lee, "Simultaneous Wireless Information and Power Transfer for Cooperative Relay Networks with Battery," *IEEE Access*, vol. 5, pp. 13171–13178,

2017, doi: 10.1109/ACCESS.2017.2724638.

- [106] H. Lee, C. Song, S. H. Choi, and I. Lee, "Outage probability analysis and power splitter designs for SWIPT relaying systems with direct link," *IEEE Commun. Lett.*, vol. 21, no. 3, pp. 648–651, 2017, doi: 10.1109/LCOMM.2016.2627055.
- [107] X. Di, K. Xiong, P. Fan, and H. C. Yang, "Simultaneous Wireless Information and Power Transfer in Cooperative Relay Networks with Rateless Codes," *IEEE Trans. Veh. Technol.*, vol. 66, no. 4, pp. 2981–2996, Apr. 2017, doi: 10.1109/TVT.2016.2588441.
- [108] S. Zhong, H. Huang, and R. Li, "Outage probability of power splitting SWIPT two-way relay networks in Nakagami-m fading," *Eurasip J. Wirel. Commun. Netw.*, vol. 2018, no. 1, Dec. 2018, doi: 10.1186/s13638-017-1006-0.
- [109] X. Bai, J. Shao, J. Tian, and L. Shi, "Power-Splitting Scheme for Nonlinear Energy Harvesting AF Relaying with Direct Link," *Wirel. Commun. Mob. Comput.*, vol. 2018, 2018, doi: 10.1155/2018/7906957.
- [110] T. S. Nguyen, N. M. Duc, H. S. Nguyen, and M. Voznak, "Outage Probability Analysis in Multiple-Relay Networks for Wireless Energy Harvesting," in 2018 41st International Conference on Telecommunications and Signal Processing, TSP 2018, Aug. 2018, doi: 10.1109/TSP.2018.8441296.
- Y. Liu, "Wireless Information and Power Transfer for Multirelay-Assisted Cooperative Communication," *IEEE Commun. Lett.*, vol. 20, no. 4, pp. 784–787, 2016, doi: 10.1109/LCOMM.2016.2535114.
- [112] H. D. Vinh, V. Van Son, and P. T. Hiep, "Performance analysis of RF Energy Harvesting Cooperative Communication Networks with DF scheme," in *Proceedings - 2019 19th International Symposium on Communications and Information Technologies, ISCIT 2019*, 2019, pp. 458–463, doi: 10.1109/ISCIT.2019.8905185.
- [113] T. D. P. Perera and D. N. K. Jayakody, "Analysis of time-switching and power-splitting protocols in wireless-powered cooperative communication system," *Phys. Commun.*, vol. 31, pp. 141–151, 2018, doi: 10.1016/j.phycom.2018.09.007.
- [114] A. Ikhlef and M. Z. Bocus, "Outage performance analysis of relay selection in SWIPT systems," in *IEEE Wireless Communications and Networking Conference, WCNC*, 2016, vol. 2016-Septe, pp. 1– 5, doi: 10.1109/WCNC.2016.7565081.
- [115] T. Li, P. Fan, and K. Ben Letaief, "Outage probability of energy harvesting relay-aided cooperative networks over rayleigh fading channel," *IEEE Trans. Veh. Technol.*, vol. 65, no. 2, pp. 972–978, Feb. 2016, doi: 10.1109/TVT.2015.2402274.
- [116] K. Zhu and F. Wang, "Relay selection in energy

harvesting cooperative networks with rateless codes," in *AIP Conference Proceedings*, Apr. 2018, vol. 1955, no. 1, p. 040109, doi: 10.1063/1.5033773.

- [117] H. Xia, Y. Li, and Y. Lu, "Relay selection optimization for SWIPT-enabled cooperative networks," *Inf.*, vol. 11, no. 1, 2020, doi: 10.3390/info11010007.
- [118] I. Krikidis, "Relay Selection in Wireless Powered Cooperative Networks with Energy Storage," *IEEE J. Sel. Areas Commun.*, vol. 33, no. 12, pp. 2596–2610, Dec. 2015, doi: 10.1109/JSAC.2015.2479015.
- [119] D. Wang, R. Zhang, X. Cheng, L. Yang, and C. Chen, "Relay selection in full-duplex energyharvesting two-way relay networks," in *IEEE Transactions on Green Communications and Networking*, 2017, vol. 1, no. 2, pp. 182–191, doi: 10.1109/TGCN.2017.2686325.
- [120] E. Ali, M. Ismail, R. Nordin, and N. F. Abdulah, "Beamforming techniques for massive MIMO systems in 5G: overview, classification, and trends for future research," *Frontiers of Information Technology and Electronic Engineering*, vol. 18, no. 6. Zhejiang University, pp. 753–772, Jun. 01, 2017, doi: 10.1631/FITEE.1601817.
- [121] D. M. Mughal, S. T. Shah, D. Munir, K. H. Park, and M. Y. Chung, "Throughput analysis of multiantenna based SWIPT relay network," in *International Conference on Information Networking*, 2018, vol. 2018-Janua, pp. 336–340, doi: 10.1109/ICOIN.2018.8343136.
- [122] X. Song, Y. Ni, X. Han, L. Qin, and L. Dong, "Time and Power Allocation for Energy Efficiency Maximization in Wireless-Powered Full-Duplex Relay Systems," vol. 11, no. 10, pp. 1–18, 2019, doi: 10.3390/fi11100205.
- Z. Zhou, M. Peng, Z. Zhao, and Y. Li, "Joint power splitting and antenna selection in energy harvesting relay channels," *IEEE Signal Process. Lett.*, vol. 22, no. 7, pp. 823–827, 2015, doi: 10.1109/LSP.2014.2369748.
- [124] Y. Liu, "Joint Resource Allocation in SWIPT-Based Multiantenna Decode-and-Forward Relay Networks," *IEEE Trans. Veh. Technol.*, vol. 66, no. 10, pp. 9192–9200, Oct. 2017, doi: 10.1109/TVT.2017.2717018.
- [125] T. Srivantana and K. Maichalernnukul, "Two-Way Multi-Antenna Relaying with Simultaneous Wireless Information and Power Transfer," *Symmetry (Basel).*, vol. 9, no. 3, p. 42, Mar. 2017, doi: 10.3390/sym9030042.
- [126] A. Albaaj, S. V. A. Makki, Q. Alabkhat, and A. Zahedi, "Simultaneous energy harvesting and information processing in wireless multiple relays with multiple antennas," *Int. J. Electron.*, vol. 104, no. 7, pp. 1214–1227, Jul. 2017, doi:

10.1080/00207217.2017.1292557.

- [127] A. Zahedi, M. Lari, A. Albaaj, and Q. Alabkhat, "Simultaneous energy harvesting and information processing considering multi-relay multi-antenna using maximum ratio transmission and antenna selection strategies," *Trans. Emerg. Telecommun. Technol.*, vol. 28, no. 11, Nov. 2017, doi: 10.1002/ett.3182.
- [128] X. Liu, Z. Wen, D. Liu, J. Zou, and S. Li, "Joint Source and Relay Beamforming Design in Wireless Multi-Hop Sensor Networks with SWIPT," *Sensors*, vol. 19, no. 1, p. 182, Jan. 2019, doi: 10.3390/s19010182.
- [129] J. Zhang and G. Pan, "Outage Analysis of Wireless-Powered Relaying MIMO Systems with Non-Linear Energy Harvesters and Imperfect CSI," *IEEE Access*, vol. 4, pp. 7046–7053, 2016, doi: 10.1109/ACCESS.2016.2617893.
- [130] K. Xu, M. Zhang, J. Liu, N. Sha, W. Xie, and L. Chen, "SWIPT in mMIMO system with non-linear energy-harvesting terminals: protocol design and performance optimization," *Eurasip J. Wirel. Commun. Netw.*, vol. 2019, no. 1, Dec. 2019, doi: 10.1186/s13638-019-1378-4.
- [131] C. Studer, M. Wenk, and A. Burg, "MIMO transmission with residual transmit-RF impairments," in 2010 International ITG Workshop on Smart Antennas, WSA 2010, 2010, pp. 189–196, doi: 10.1109/WSA.2010.5456453.
- [132] P. Handel and D. Ronnow, "Dirty MIMO transmitters: Does it matter?," *IEEE Trans. Wirel. Commun.*, vol. 17, no. 8, pp. 5425–5436, Aug. 2018, doi: 10.1109/TWC.2018.2843764.
- [133] S. Cheng, R. Wang, J. Wu, W. Zhang, and Z. Fang, "Performance Analysis and Beamforming Designs of MIMO AF Relaying with Hardware Impairments," *IEEE Trans. Veh. Technol.*, vol. 67, no. 7, pp. 6229–6243, Jul. 2018, doi: 10.1109/TVT.2018.2817582.
- [134] Z. Liu, G. Lu, Y. Ye, and X. Chu, "System Outage Probability of PS-SWIPT Enabled Two-Way AF Relaying with Hardware Impairments," *IEEE Trans. Veh. Technol.*, vol. 69, no. 11, pp. 13532– 13545, Nov. 2020, doi: 10.1109/TVT.2020.3030134.
- [135] X. Li *et al.*, "Joint effects of residual hardware impairments and channel estimation errors on swipt assisted cooperative noma networks," *IEEE Access*, vol. 7, pp. 135499–135513, 2019, doi: 10.1109/ACCESS.2019.2942337.
- [136] X. Li, J. Li, P. Takis Mathiopoulos, D. Zhang, L. Li, and J. Jin, "Joint Impact of Hardware Impairments and Imperfect CSI on Cooperative SWIPT NOMA Multi-Relaying Systems," in 2018 IEEE/CIC International Conference on Communications in China, ICCC 2018, Feb. 2019, pp. 95–99, doi: 10.1109/ICCChina.2018.8641194.

- [137] X. Li, J. Li, and L. Li, "Performance Analysis of Impaired SWIPT NOMA Relaying Networks over Imperfect Weibull Channels," *IEEE Syst. J.*, vol. 14, no. 1, pp. 669–672, Mar. 2020, doi: 10.1109/JSYST.2019.2919654.
- [138] K. Wang, Y. Li, Y. Ye, and H. Zhang, "Dynamic power splitting schemes for non-linear EH relaying networks: Perfect and imperfect CSI," in *IEEE Vehicular Technology Conference*, Feb. 2018, vol. 2017-September, pp. 1–5, doi: 10.1109/VTCFall.2017.8287945.
- [139] D. Zhai, R. Zhang, J. Du, Z. Ding, and F. R. Yu, "Simultaneous wireless information and power transfer at 5G new frequencies: Channel measurement and network design," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 1, pp. 171–186, Jan. 2019, doi: 10.1109/JSAC.2018.2872366.
- [140] N. Javaid, A. Sher, H. Nasir, and N. Guizani, "Intelligence in IoT-Based 5G Networks: Opportunities and Challenges," *IEEE Commun. Mag.*, vol. 56, no. 10, pp. 94–100, 2018, doi: 10.1109/MCOM.2018.1800036.
- [141] L. Da Xu, W. He, and S. Li, "Internet of things in industries: A survey," *IEEE Transactions on Industrial Informatics*, vol. 10, no. 4. IEEE Computer Society, pp. 2233–2243, Nov. 01, 2014, doi: 10.1109/TII.2014.2300753.
- [142] D. Ciuonzo, P. S. Rossi, and S. Dey, "Massive MIMO channel-aware decision fusion," *IEEE Trans. Signal Process.*, vol. 63, no. 3, pp. 604–619, 2015, doi: 10.1109/TSP.2014.2376886.
- [143] S. S. Adhatarao, M. Arumaithurai, D. Kutscher, and X. Fu, "ISI: Integrate Sensor Networks to Internet With ICN," *IEEE Internet Things J.*, vol. 5, no. 2, pp. 491–499, Apr. 2018, doi: 10.1109/JIOT.2017.2741923.
- [144] W. Guo, S. Zhou, Y. Chen, S. Wang, X. Chu, and Z. Niu, "Simultaneous Information and Energy Flow for IoT Relay Systems with Crowd Harvesting," *IEEE Commun. Mag.*, vol. 54, no. 11, pp. 143–149, 2016, doi: 10.1109/MCOM.2016.1500649CM.
- [145] M. Amjad, A. Ahmed, M. Naeem, M. Awais, W. Ejaz, and A. Anpalagan, "Resource Management in Energy Harvesting Cooperative IoT Network under QoS Constraints," *Sensors*, vol. 18, no. 10, p. 3560, Oct. 2018, doi: 10.3390/s18103560.
- [146] W. Lu, G. Liu, P. Si, G. Zhang, B. Li, and H. Peng, "Joint resource optimization in simultaneous wireless information and power transfer (SWIPT) enabled multi-relay internet of things (IoT) system," *Sensors (Switzerland)*, vol. 19, no. 11, p. 2536, 2019, doi: 10.3390/s19112536.
- [147] H. Kawabata, K. Ishibashi, S. Vuppala, and G. T. F. De Abreu, "Robust Relay Selection for Large-Scale Energy-Harvesting IoT Networks," *IEEE Internet Things J.*, vol. 4, no. 2, pp. 384–392, 2017, doi:

10.1109/JIOT.2016.2586581.

- [148] Y. Hu, Y. Zhu, M. C. Gursoy, and A. Schmeink, "SWIPT-Enabled relaying in IoT networks operating with finite blocklength codes," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 1, pp. 74–88, 2019, doi: 10.1109/JSAC.2018.2872361.
- [149] Y. Zou, J. Zhu, and X. Jiang, "Joint Power Splitting and Relay Selection in Energy-Harvesting Communications for IoT Networks," *IEEE Internet Things J.*, vol. 7, no. 1, pp. 584–597, Jan. 2020, doi: 10.1109/JIOT.2019.2946651.
- [150] D. K. P. Asiedu, H. Lee, and K. J. Lee, "Simultaneous Wireless Information and Power Transfer for Decode-and-Forward Multihop Relay Systems in Energy-Constrained IoT Networks," *IEEE Internet Things J.*, vol. 6, no. 6, pp. 9413– 9426, Dec. 2019, doi: 10.1109/JIOT.2019.2937090.
- [151] X. Li *et al.*, "Cooperative Wireless-Powered NOMA Relaying for B5G IoT Networks with Hardware Impairments and Channel Estimation Errors," *IEEE Internet Things J.*, pp. 1–1, Oct. 2020, doi: 10.1109/jiot.2020.3029754.
- [152] C.-B. Le, D.-T. Do, and M. Voznak, "Exploiting Impact of Hardware Impairments in NOMA: Adaptive Transmission Mode in FD/HD and Application in Internet-of-Things," *Sensors*, vol. 19, no. 6, p. 1293, Mar. 2019, doi: 10.3390/s19061293.
- [153] H. Guo, X. Guo, C. Deng, and S. Zhao, "Performance Analysis of IQI Impaired Cooperative NOMA for 5G-Enabled Internet of Things," *Wirel. Commun. Mob. Comput.*, vol. 2020, 2020, doi: 10.1155/2020/3812826.
- [154] H. Sun, F. Zhou, R. Q. Hu, and L. Hanzo, "Robust beamforming design in a NOMA cognitive radio network relying on SWIPT," *IEEE J. Sel. Areas Commun.*, vol. 37, no. 1, pp. 142–155, Jan. 2019, doi: 10.1109/JSAC.2018.2872375.
- [155] X. Li, M. Huang, Y. Liu, V. G. Menon, A. Paul, and Z. Ding, "I/Q Imbalance Aware Nonlinear Wireless-Powered Relaying of B5G Networks: Security and Reliability Analysis," *arXiv*. arXiv, Jun. 06, 2020, doi: 10.1109/tnse.2020.3020950.
- [156] A. Rauniyar, P. E. Engelstad, and O. N. Osterbo, "Performance analysis of rf energy harvesting and information transmission based on noma with interfering signal for iot relay systems," *IEEE Sens. J.*, vol. 19, no. 17, pp. 7668–7682, Sep. 2019, doi: 10.1109/JSEN.2019.2914796.
- [157] S. A. Khan, A. Kavak, sultan Aldirmaz colak, and K. Kucuk, "A Novel Fractional Frequency Reuse Scheme for Interference Management in LTE-A HetNets," *IEEE Access*, 2019, doi: 10.1109/access.2019.2933689.
- [158] S. A. Khan, M. Asshad, A. Kavak, and K. Küçük, "A Novel Radio Resource Management Technique for Femtocells in LTE-Pro Networks," in

International Engineering Research Symposium (UMAS 2017, 2017, pp. 1–4.

- [159] S. A. Khan, A. Kavak, K. Küçük, and M. Asshad, "A New Fractional Frequency Reuse Method for Interference Management in LTE-A HetNets," in 27th Signal Processing and Communications Applications Conference, SIU 2019, 2019, doi: 10.1109/SIU.2019.8806317.
- [160] J. Zhang, E. Bjornson, M. Matthaiou, D. W. K. Ng, H. Yang, and D. J. Love, "Prospective Multiple Antenna Technologies for beyond 5G," *IEEE J. Sel. Areas Commun.*, vol. 38, no. 8, pp. 1637–1660, Aug. 2020, doi: 10.1109/JSAC.2020.3000826.
- [161] J. Zhang, S. Chen, Y. Lin, J. Zheng, B. Ai, and L. Hanzo, "Cell-free massive MIMO: A new nextgeneration paradigm," *IEEE Access*, vol. 7, pp. 99878–99888, 2019, doi: 10.1109/ACCESS.2019.2930208.
- [162] X. Chen, D. W. K. Ng, W. Yu, E. G. Larsson, N. Al-Dhahir, and R. Schober, "Massive access for 5g and beyond," *arXiv*. arXiv, Feb. 09, 2020, doi: 10.1109/jsac.2020.3019724.
- [163] X. Chen, D. W. K. Ng, and H. H. Chen, "Secrecy wireless information and power transfer: Challenges and opportunities," *IEEE Wirel. Commun.*, vol. 23, no. 2, pp. 54–61, 2016, doi: 10.1109/MWC.2016.7462485.
- [164] M. Asshad, S. A. Khan, A. Kavak, K. Küçük, and D. L. Msongaleli, "Cooperative Communications using Relay Nodes for Next-Generation Wireless Networks with Optimal Selection Techniques: A Review," *IEEJ Trans. Electr. Electron. Eng.*, vol. 14, no. 5, 2019, doi: 10.1002/tee.22852.
- [165] Y. Liang, Y. He, J. Qiao, and A. P. Hu, "Simultaneous Wireless Information and Power Transfer in 5G Mobile Networks: A Survey," in 2019 Computing, Communications and IoT Applications, ComComAp 2019, Oct. 2019, pp. 460–465, doi: 10.1100/ComComAp.2010.0018827

10.1109/ComComAp46287.2019.9018837.

- [166] S. Ladan, A. B. Guntupalli, and K. Wu, "A highefficiency 24 GHz rectenna development towards millimeter-wave energy harvesting and wireless power transmission," *IEEE Trans. Circuits Syst. I Regul. Pap.*, vol. 61, no. 12, pp. 3358–3366, 2014, doi: 10.1109/TCSI.2014.2338616.
- [167] W. Hao et al., "Beamforming Design in SWIPT-Based Joint Multicast-Unicast mmWave Massive MIMO with Lens-Antenna Array," *IEEE Wirel. Commun. Lett.*, vol. 8, no. 4, pp. 1124–1128, 2019, doi: 10.1109/LWC.2019.2908829.
- [168] G. N. Kamga and S. Aissa, "Wireless power transfer in mmWave massive MIMO systems with/without rain attenuation," *IEEE Trans. Commun.*, vol. 67, no. 1, pp. 176–189, Jan. 2019, doi: 10.1109/TCOMM.2018.2799217.
- [169] G. N. Kamga and S. Aissa, "Relay-Aided Energy

This article has been accepted for publication in a future issue of this journal, but has not been fully edited. Content may change prior to final publication. Citation information: DOI 10.1109/ACCESS.2021.3078703, IEEE Access



and Information Transmission in mmWave Massive MIMO Systems," in 2017 IEEE Global Communications Conference, GLOBECOM 2017 - Proceedings, Jul. 2017, vol. 2018-January, pp. 1–6, doi: 10.1109/GLOCOM.2017.8254956.

- [170] S. Biswas, S. Vuppala, and T. Ratnarajah, "On the Performance of mmWave Networks Aided by Wirelessly Powered Relays," *IEEE J. Sel. Top. Signal Process.*, vol. 10, no. 8, pp. 1522–1537, Dec. 2016, doi: 10.1109/JSTSP.2016.2610944.
- [171] M. Xu and L. Liu, "Sender-Receiver Role-Based Energy-Aware Scheduling for Internet of Underwater Things," *IEEE Trans. Emerg. Top. Comput.*, vol. 7, no. 2, pp. 324–336, Apr. 2019, doi: 10.1109/TETC.2016.2632749.
- [172] M. Jahanbakht, W. Xiang, L. Hanzo, and M. R. Azghadi, "Internet of Underwater Things and Big Marine Data Analytics—A Comprehensive Survey," *IEEE Commun. Surv. Tutorials*, 2021, doi: 10.1109/COMST.2021.3053118.
- [173] T. Qiu, Z. Zhao, T. Zhang, C. Chen, and C. L. Philip Chen, "Underwater Internet of Things in Smart Ocean: System Architecture and Open Issues," *IEEE Trans. Ind. Informatics*, vol. 16, no. 7, pp. 4297–4307, Jul. 2020, doi: 10.1109/TII.2019.2946618.

Nausheen Ashraf received her B.Sc degree in Electrical Engineering from the University of Engineering & Technology, Lahore in 2007. She completed her Master's degree in Electrical Engineering from the University of Engineering & Technology Lahore in 2013. She is currently a Ph.D. student at the Department of Electrical Engineering, College of Electrical & Mechanical Engineering (CEME), National University of Science and Technology (NUST), Islamabad Pakistan. Her main research interests include 5G networks, cooperative communication, mm-wave, wireless power transfer, and the internet of things (IoT).



Shahzad Amin Sheikh received B.S. degree in electrical engineering from Eastern Mediterranean University (EMU), Famagusta, North Cyprus, in 1995 and the Master's degree in electrical engineering from the University of New South Wales (UNSW), Sydney, N.S.W., Australia in

2002. He received his Ph.D. in electrical engineering from Southwest Jiaotong University, Chengdu, China, in 2008. He is currently a Professor with the Department of Electrical Engineering, College of Electrical and Mechanical Engineering, National University of Science and Technology (NUST), Islamabad, Pakistan. His current research interests include machine learning, internet of things (IoT), signal processing and its applications in digital communication.



Sajjad Ahmad Khan received his B.Sc. degree in Information Technology from the University of Engineering and Technology (UET), Peshawar, Pakistan, and M.Sc. degree in Telecommunication and Networking from Iqra University, Islamabad, Pakistan in 2008 and 2011,

respectively. He completed his Ph.D. degree in Computer Engineering from Kocaeli University, Turkey in 2020. He is currently a post-doc research fellow at the Electronics and Communication Engineering Department, Istanbul Technical University (ITU), Turkey. He was also a visiting researcher at Adam Mickiewicz University, Poznań, Poland. He is an active reviewer of many conferences and journals. His main research interests include 5G, and beyond Wireless Communication, Mobility and Handover Management, Cooperative Communication, Radio Resource Management, and Internet Of Things (IoT).



Ibraheem Shayea (M'15) received the B.Sc. degree in electronic engineering from the University of Diyala, Baqubah, Iraq, in 2004, and the M.Sc. degree in computer and communication engineering and the Ph.D. degree in mobile communication engineering

from the National University of Malaysia, University Kebangsaan Malaysia (UKM), Malaysia, in 2010 and 2015, respectively. From January 2011 to February 2014, he worked as a Research and Teaching Assistant with UKM. Then, from January 2016 to June 2018, he joined the Wireless Communication Center (WCC), University of Technology Malaysia (UTM), Malaysia, and worked there as a Research Fellow. He has been a Researcher Fellow with Istanbul Technical University (ITU), Istanbul, Turkey, since September 2018. His main research interests include wireless communication systems, mobility management, radio propagation, and the Internet of Things (IoT).

Marium Jalal received her Post Doc & PhD in Electrical Engineering from École de technologie supérieure ÉTS, Montreal Canada & University of L'Aquila, L'Aquila Italy respectively. She was Awarded with Gold Medal for the best M.Sc. Thesis. She has been a Marie-Curie Early Stage Researcher with the Center of Excellence for Research DEWS, University of L'Aquila, with WEST Aquila s.r.l., Italy and with ActiCom Berlin, Germany. She is recipient of several international & national research and travel grant awards. Dr. Maium is a Pakistan Engineering Council (PEC) Approved Program Evaluator / Subject Expert in the field of Electrical/ Electronic Engineering. She is currently working as Head of Electronic Engineering Department at Fatima Jinnah Women University, Rawalpindi.